# **Advanced Graphics Package**

Version 2.0

# Supplement for HP 1000 Systems





HEWLETT - PACKARD COMPANY Engineering Productivity Division 19420 Homestead Road Cupertino, California 95014 MANUAL PART NO. 92862-90001 Printed in U.S.A. May 1984 E0584

## **PRINTING HISTORY**

The Printing History below identifies the Edition of this Manual and any Updates that are included. Periodically, Update packages are distributed which contain replacement pages to be merged into the manual, including an updated copy of this Printing History page. Also, the update may contain write-in instructions.

Each reprinting of this manual will incorporate all past Updates. However, no new information will be added. Thus, the reprinted copy will be identical in content to prior printings of the same edition with its user-inserted update information. New editions of this manual will contain new information, as well as all Updates.

First Edition May 1984

## NOTICE

The information contained in this document is subject to change without notice.

HEWLETT-PACKARD MAKES NO WARRANTY OF ANY KIND WITH REGARD TO THIS MATERIAL, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE. Hewlett-Packard shall not be liable for errors contained herein or for incidental consequential damages in connection with the furnishing, performance or use of this material.

Hewlett-Packard assumes no responsibility for the use or reliability of its software on equipment that is not furnished by Hewlett-Packard.

This document contains proprietary information which is protected by copyright. All rights are reserved. No part of this document may be photocopied, reproduced or translated to another language without the prior written consent of Hewlett-Packard Company.

#### Copyright (c) 1984 by HEWLETT-PACKARD COMPANY

## PREFACE

#### Who needs to use this manual?

This Supplement complements the AGP Reference Manual and is intended for users of Hewlett-Packard's Advanced Graphics Package (AGP). The Supplement presents AGP reference information which is specific to the HP 1000 operating systems.

#### How is it organized?

There are four parts to this Supplement plus an Appendix. Part I contains HP 1000 Operating System topics such as programming requirements, usermodifiable graphics tables, and other HP 1000 system-specific topics. Part II describes the AGP programs (User Program, Work Station Program and Monitor Program). Part III describes calls which are operating system dependent. Part IV describes AGP installation procedures and Appendix A summarizes the changes in the AGP Subroutine Parameters from Version 1.0.

# HP Computer Museum www.hpmuseum.net

For research and education purposes only.

# Table of Contents

## Part I

## System-Dependent Topics

Operating EnvironmentI-1
Logical UnitsI-1
Programming LanguagesI-1
FORTRAN
Pascal
MACRO/1000I-9
Naming ConventionsI-11
Library File NamesI-11
Linking AGP ProgramsI-13
The User Program
The Work Station ProgramI-13
EMA/VMA Device HandlersI-15
The Monitor Program
Large User Programs (Non-CDS Environment)I-16
Spooling
General
Device Output
File Output
Using Graphics Terminals
Using HP Plotters and Translators
using me recovers and managedors

## Part II

## Advanced Topics

Description of AGP ProgramsII-1
User ProgramII-1
Work Station ProgramII-1
Monitor ProgramII-2
Abnormal Program TerminationII-3
Concurrent WSP SchedulingII-3
AGP Program-to-Program CommunicationsII-4
Class I/O Allocation and De-allocation
The Segment Display AreaII-5
Segment Display Area OverflowII-6
Changing the SizeII-6
Linking a WSP Without a Segment Display Area
The Segment Hashing TableII-8
Changing the SizeII-9
Polygon Set Effeciency ConsiderationsII-10
Changing the Size of the Polygon Style Table
Changing the Size of the Intercept Buffer
Changing the Size of the Vertex Buffer

v

Loading Dummy Polygon Set Modules Color Modelling Efficiency Considerations Changing the Size of the Color Table Loading Dummy Color Modelling Routines Fonts	II-15 II-15 II-16
Part III System-Dependent Calls	
IntroductionJDFNTJDINT	
JEDEVJIWSJSDFJSERR	
Part IV AGP Installation	
System Prerequisites for Installation Installation Procedure	
Appendix A AGP Subroutine Changes	
Summary of AGP Subroutine Changes Specific Changes	

# List of Tables

AGP and	Pascal D	)ata	Type C	orrespondenceI-6	б
Device H	Handler F	lile	Naming	ConventionsI-12	2

## AGP SUBROUTINES AND PARAMETERS FOR HP 1000 SYSTEMS

J2DRW (X,Y) J2MOV (X,Y) J2MRK (X,Y,MARKNO) J2PGN (NPOLY,NVEC,XVEC,YVEC) J2PLY (NPNTS,XVEC,YVEC) J3DRW (X,Y,Z) J3MOV (X,Y,Z) J3MRK (X,Y,Z,MARKNO) J 3PGN (NPOLY, NVEC, XVEC, YVEC, ZVEC) J3PLY (NPNTS, XVEC, YVEC, ZVEC) JALPH (ID,NCHARS,STRING) JASPK (XSIZE, YSIZE) JBATC {no parameters} JBEGN {no parameters} JBUTN (ID, ECHO, BUTTON) JCLOS {no parameters} JCLPD (HITHR, YON) JCLPW (OPCODE) JCLR {no parameters} JCMOD (OPCODE) JCOLM (ID, MODEL) JCOLR ( COLOR ) JCORI (XBAS, YBAS, ZBAS, XPLN, YPLN, ZPLN) JCSIZ (WIDTH, HEIGHT, GAP) JDCOL (ID,COLOR,COLP 1,COLP 2,COLP 3) JDDEV (ID.CLASS) JDFNT (FONT, SLANT, FNLEN, FNAME, CONTRL) JDINT (ID, WSPLEN, WSPNAM, LU, CONTRL) JDLIM (ID,XMIN,XMAX,YMIN,YMAX) JDMOD (MATRIX) JDPMM (ID,VX,VY,XMM,YMM) JDPST (ID, PINDEX, DENSTY, ORIENT, EDGE, DIDDF) JDPTH (HDIST, YDIST) JEDEV (ID.CLASS.LU) JEND {no parameters} JFONT (FONT) JGDET (OPCODE) JGHI (OPCODE) JGVIS (OPCODE) JHAND (OPCODE) JI 1IN (OPCODE, VALUE) JI 1RE (OPCODE, VALUE) JI2RE (OPCODE, VAL 1, VAL 2) JI 3RE (OPCODE, VAL 1, VAL 2, VAL 3) JI4RE (OPCODE, VAL 1, VAL 2, VAL 3, VAL 4) JICOL (ID,COLOR,COLP 1,COLP 2,COLP 3) JICP (X,Y,Z) JIERR (ERRNUM.LEVNUM.SUBNUM.INFO) JIESC (ID, OPCODE, iSIZE, RSIZE, ILIST, RLIST) JIMAT (OPCODE, MATRIX) JIPST (ID,INDEX,DENSTY,ORIENT,EDGE,DIDDF) JISGA (NAME, VALUE) JISGW (SEGNAM.MAXNAM.COUNT.IDARRY) JITSZ (NCHARS, STRING, SHIGH, SWIDE) JIVOF (no parameters) JIVON {no parameters} JIWND (XARRAY, YARRAY, ZARRAY)

JIWS (ID, OPCODE, ISIZE, RSIZE, ILIST, RLIST) JJUST (BASE,UP) JKYBD (ID,ECHO,MAX,ACTUAL,STRING) JLLIM (ID,XMIN,XMAX,YMIN,YMAX) JLOCP (ID,VX,VY) JLPMM (ID,VX,VY,XMM,YMM) JLSTL (LSTYLE) JLWID (LWIDTH) JMCUR {no parameters} JNEWF (no parameters) JOESC (ID, OPCODE, ISIZE, RSIZE, ILIST, RLIST) JOPEN (NAME) JPICK (ID, ECHO, BUTTON, NAME, PICKID) JPICL (COLOR) JPIKP (ID,VX,VY) JPILS (LSTYLE) JPKID (PICKID) JPLIM (ID,XMIN,XMAX,YMIN,YMAX) JPPMM (ID,VX,VY,XMM,YMM) JPROJ (OPCODE, DU, DV, DN) JPSTL (PINDEX) JPURG (NAME) JR2DR (DX.DY) JR2MK (DX,DY,MARKNO) JR2MV (DX,DY) JR2PG (NPOLY, NVEC, DXVEC, DYVEC) JR 2PL (NPNTS, DXVEC, DYVEC) JR 3DR (DX,DY,DZ) JR 3MK (DX,DY,DZ,MARKNO) JR 3MV (DX,DY,DZ) JR 3PG (NPOLY, NVEC, DXVEC, DYVEC, DZVEC) JR 3PL (NPNTS, DXVEC, DYVEC, DZVEC) JRNAM (OLDNAM, NEWNAM) JRSET (no parameters) JSDET (NAME, OPCODE) JSDF (ID,NAMLEN,NAME,SIZE,CONTRL) JSERR (RLU,RLEVEL,ALEVEL) JSHI (NAME,OPCODE) JSLOC (ID,ECHO,VX,VY) JSVAL (ID, ECHO, SUBVAL, VALUE) JSVIS (NAME, OPCODE) JTEXH (NCHARS, STRING) JTEXL (NCHARS, STRING) JTEXM (NCHARS, STRING) JUPDT {no parameters} JVDIS (DIST) JVIEW (VPXMIN, VPXMAX, VPYMIN, VPYMAX) JVPLN (XNRM, YNRM, ZNRM, XUP, YUP, ZUP) JVREF (X,Y,Z) JVSAL (OPCODE) JVTOW (VX,VY,X,Y,Z) JWEND (ID) JWIND (DWUMIN, DWUMAX, DWVMIN, DWVMAX) JWLOC (ID,ECHO,BUTTON,VX,VY) JWOFF (ID) JWON (ID) JWTOV (X,Y,Z,VX,VY) JWVAL (ID,ECHO, SUBVAL, BUTTON, VALUE)

· | |

# Part I System-Dependent Topics

## OPERATING ENVIRONMENT

AGP is supported on the following HP 1000 processors and RTE operating systems:

Processor	Operating System
HP 1000 E,F	RTE-6/VM
HP 1000 A600,A700,A900	RTE-A

RTE-A Virtual Code (VC+)

Note: Any reference within this manual to RTE-A applies equally to RTE-A with VC+ unless otherwise stated. The term 'CDS' in this manual refers to Code and Data Separation, available exclusively under the RTE-A Operating System with the VC+ System Extension Package. CDS is not available under any other HP operating system, and the programs compiled for other systems should always be non-CDS.

## LOGICAL UNITS

DGL supports a logical unit number (LU) in the range of 0 to 63. For information regarding LUs, refer to the RTE-A User's Manual, the RTE-A Utilities Manual or the RTE-6/VM CI User's Manual.

## PROGRAMMING LANGUAGES

AGP is designed to have a multi-language interface. This means, it is possible for the user to access AGP from different programming languages using identical functions. The multi-language interface can be called from FORTRAN, Pascal and MACRO.

## FORTRAN

Application programs written in FORTRAN 7X must pass parameters to AGP in the form AGP requires. All integers must be 16-bit integers. All reals must be 32-bit reals.

Characters must be passed to AGP and are returned from AGP in integer arrays. The following example illustrates how characters may be used in an integer array:

> INTEGER TEXT(8) DATA TEXT/2HAB,2HCD,2HEF,2HGH/

All parameters passed to AGP must have one word addresses and must not reside in EMA/VMA.

The programming examples of this manual are written in FORTRAN, however, they are not complete programs. Care should be exercised when combining them into programs which are to be executed, because many calls which must be used (but are not related to the topic being presented) are not shown.

The FORTRAN example shown on the following page is complete. It may be typed into the computer and executed, or it may be modified to include many of the other program fragments.

```
С
C..Delete the C in column 1 of the following CDS directive when
C linking with the CDS version of UPLIB in the VC+ environment.
C$CDS ON
     PROGRAM USER
С
C..AGP User Program to draw a line.
C...Set up variables.
C
     INTEGER WSPLEN, WSPNAM(4), LU, CNTRL
            WSPLEN /8/
     DATA
            WSPNAM /2HWS,2HP1,2H.R,2HUN/
     DATA
С
C.. Initialize the graphics system.
С
     CALL JBEGN
С
C...Initialize a work station program: work station 1,WSPNAM =
C 'WSP1.RUN' at LU = 1 with no control bits set.
C Enable it for graphics output.
С
     LU
          = 1
     CNTRL = 0
     CALL JDINT(1, WSPLEN, WSPNAM, LU, CNTRL)
     CALL JWON(1)
С
C.. Everything between the comment lines may be changed as needed.
С
C... Use default attributes and viewing transformation to draw a line
C diagonally across the display.
С
     CALL J2MOV(-1.0, -1.0)
     CALL J2DRW( 1.0, 1.0)
С
С
C..Disable the work station, terminate it, end the AGP system, and
C terminate this program.
C
     CALL JWOFF(1)
     CALL JWEND(1)
     CALL JEND
С
     END
```

## Pascal

In general, the considerations which must be made when accessing AGP routines are the same as when accessing any non-Pascal routine from a Pascal program. Therefore, a review of that topic in the *Pascal Language Manual* may be useful.

Parameters (param1, param2, ...) passed to AGP routines must be declared VAR, or call-by-reference. Note that any external routine may modify the actual parameters passed to it when declared VAR in Pascal. However, AGP does not modify input parameters. All AGP routines should be declared as external procedures as follows:

PROCEDURE AGP\_routine
\$ALIAS 'Jxxxx'\$
(VAR param1 : type1;
VAR param2 : type2;
:
:
VAR paramN : typeN);
EXTERNAL;

where Jxxxx is the true name of the AGP routine, and AGP\_routine is a convenient alias.

The following files are supplied with AGP:

PAGP1.PASI PAGP2.PASI PAGP3.PASI

These files provide standard data types and external declarations for inclusion in all application programs. The contents of these files may be copied to a local directory and modified as needed.

The application program must use the data types and format which AGP requires. As a Pascal user, you should declare and use appropriate data types for both the formal and actual parameters of AGP routines. One example of a data type where considerations must be made is the data type INTEGER. The Pascal data type INTEGER represents integers in 32 bits, while AGP's code always assumes them to be 16 bits.

The following declarations cause the variable graphics\_id to be stored in a 16-bit integer:

TYPE INT = -32768..32767; VAR

graphics\_id : INT;

Characters must also be formatted appropriately. Pascal data type CHAR is stored as one character per 16-bit word, while AGP requires two characters

in 16 bits. The application program must declare all parameters containing characters to be PACKED ARRAYS of the appropriate size. If an application program were going to pass an array of 40 characters, the variable "text string" should be declared as follows:

TYPE

CHAR40 = PACKED ARRAY [1..40] of CHAR;

VAR

text string : CHAR40;

Some AGP routines pass back information through parameters which could be interpreted as different Pascal data types, depending on the use of the AGP routine. JIWS is an example of such a routine. For opcodes that inquire device information (5050, 5051, 6050, 6051, 7050 and 7051) the ILIST array parameter to the JIWS routine returns information which, for part of the array, is interpreted as characters. The remainder of the array, is interpreted as integers.

If the JIWS routine is declared from a Pascal program, as it is from a FORTRAN program, code is dedicated to the translation of the character from the integer array. This is done in Pascal by either computationally extracting the high and low order bytes of each integer to get the corresponding character, or by using a variant record structure that automatically maps the necessary integers to a packed array of characters. An example follows:

TYPE SINT = -32768..32767; [Single Word Integer] LET\_NUM = (Alpha, Numeric); AGP\_STRING = RECORD CASE LET\_NUM OF Alpha : (letters : PACKED ARRAY[1..2] OF CHAR): Numeric : (num : SINT); END; END;

Using this example, each element in the ILIST array is of the type AGP\_STRING (i.e., VAR ILIST) : ARRAY [1..6] OF AGP\_STRING;). If the first element of the array passed back by JIWS contained characters, they could be accessed in the following manner:

first\_letter :=ILIST[1].letters[1];
second letter :=ILIST[1].letters[2];

A more straightforward solution to this mapping problem is to define a special AGP routine, aliasing to JIWS, with the mapping done transparently. The following type declarations and procedure declaration could be used for the JIWS call with OPCODES of 5050, 5051, 6050, 6051, 7050 and 7051. (Note that in this example, device\_status is the more descriptive variable name for ILIST).

TYPE = -32768..32767; {Single Word Integer} SINT DEV STATUS = RECORD name : PACKED ARRAY [1..6] OF CHAR; state : ARRAY [1..4] OF SINT; END: REAL LIST = ARRAY [1..4] OF REAL; PROCEDURE inquire device status \$ALIAS "JIWS"\$ (VAR opcode, { inquiry opcode { size of the integer array isize rsize : SINT; { size of the real array VAR device\_status : DEV STATUS; VAR dummy : REAL LIST; VAR ierr : SINT );

It is also possible (as for JOESC and JIESC) that an AGP routine could pass either character or integer information through the same parameter, depending on the desired interpretation of the call. The Pascal program could define the AGP routine in a variety of ways to gain the desired amount of flexibility. Variant records, as presented in the first solution, could also be used to achieve the desired results.

}

}

Table 1 summarizes correspondences between AGP parameters and matching Pascal data types. A file containing standard data-type declarations for Pascal is supplied with the AGP product.

AGP expects	Pascal data type
Integer	-3276832767
Character string of n characters	PACKED ARRAY[1n] OF CHAR
Real	REAL
Array of n integers	ARRAY [1n] of -3276832767
Array of n reals	ARRAY [1n] of real

 Table 1. AGP and Pascal Data Type Correspondence

Another consideration when accessing AGP from Pascal, is the type of heap used. The heap is a memory area that may be allocated at run time. Its location, relative to the program code, is determined by the HEAP compiler option. The heap may exist within the program's logical address space (HEAP 1), in Extended Memory Area (HEAP 2), or may not exist at all (HEAP 0). Only HEAP 2 requires special consideration by AGP users.

When HEAP 2 is used, the heap is in EMA. The address of format parameters are two words long. AGP routines require their parameters have a single word address. However, the program may force Pascal to use only one word for those parameter's address via another compiler option: HEAPPARMS OFF. HEAPPARMS OFF should be set before the declaration of the formal parameters of AGP routines. This tells the compiler to give them a single word address. It also prevents the actual parameters from is necessary when accessing AGP.

\$HEAP 2\$

PROGRAM agp;

TYPE INT= -32768..32767;

PROCEDURE get button
 \$ALIAS "JBUTN"\$
 (\$HEAPPARMS OFF\$
 VAR echo\_type,
 button : INT
 \$HEAPPARMS ON\$);
EXTERNAL;
 :

:

```
A Pascal version of program USER follows:
{ Delete the comment delimiters from around the $CDS ON directive
{ when linking the following program with the CDS version of UPLIB
{ in the VC+ environment.
\{\text{CDS ON}\}
PROGRAM user; {AGP user program to draw a line }
{ constant and type declarations }
{ the pagp1.inc include file is part of the AGP product }
$INCLUDE 'pagp1.pasi'$
VAR
  id, wsplen, lu, control : int;
  wspnam : string63;
  xmin, xmax, ymin, ymax : real;
{ external procedure declarations }
{ the pagp2.inc and pagp3.inc include files are part of the AGP }
{ product.
$INCLUDE 'pagp2.pasi'$
$INCLUDE 'pagp3.pasi'$
BEGIN {Main of user }
{ Initialize variables }
  id := 1;
  wsplen := 8;
  wspnam := 'WSP1.RUN';
  lu := 1;
  control := 0;
  xmin := -1.0;
  xmax := 1.0;
  ymin := -1.0;
  xmin := 1.0;
{ Initialize the graphics system }
   jbegn;
{ Initialize a work station program: work station 1, wspnam =
{ 'WSP1.RUN' at lu = 1 with no control bits set.
{ Enable it for graphics output
  jdint (id, wsplen, wspnam, lu, control);
  jwon (id);
{ Everything between the dashes may be changed as needed.
{ Use default attributes and viewing transformation to draw a
{ line diagonally across the display
   j2mov (xmin, ymin);
   j2drw (xmax, ymax);
                              { Disable the work station, terminate it, and end the AGP system.
   jwoff (id);
   jwend (id);
   jend;
END. { Main of user }
```

} }

}

## MACRO/1000

Application programs written in MACRO/1000 must pass parameters to AGP in the form AGP requires. All integers must be 16-bit integers. All reals must be 32-bit reals. Characters must be packed into contiguous 16-bit words, one character per byte, and left-justified. All parameters passed to AGP must have a one word address and must not reside in EMA/VMA.

For additional assistance, refer to the MACRO/1000 Manual section on calling subroutines and CDS programming. Two MACRO/1000 versions of the program USER, (non-CDS and CDS) are shown in the following examples:

#### Non-CDS Version:

```
MACRO, R, Q
     NAM USER
                   ;AGP User Program to draw a line
     EXT JBEGN, JDINT, JWON, J2MOV, J2DRW, JWOFF, JWEND, JEND, EXEC
٠
 ..Constants:
ID
     DEC 1
                   ;work station id
WSPLEN DEC 8
                   ;work station name length
WSPNAM ASC 8, WSP1. RUN ; work station program name
                  ;graphics output LU
    DEC 1
GLU
                  display initialization control word
CNTRL DEC 0
                  ;X and Y values of the beginning of the line
M1.0 DEC-1.0
                 ;X and Y values of the line's endpoint
D1.0 DEC 1.0
     DEC 6
D6
                  ;exec code for program termination
*..Main routine
USER NOP
                   ; initialize the graphics system
     JSB JBEGN
      DEF *+1
                   ; return address
     JSB JDINT
                   ; initialize the work station
      DEF *+6
      DEF ID
                   ; work station 1
                  ; work station name length = 8
      DEF WSPLEN
      DEF WSPNAM
                 ; name = 'WSP1.RUN'
                   ; LU of the graphics display device (LU = 1)
      DEF GLU
      DEF CNTRL
                   ; no control bits set
                   ;enable the work station for graphics output
     JSB JWON
      DEF *+2
      DEF ID
                   ; work station 1
*.. Everything between the asterisks may be changed as needed.
             *****
*.. Using default attributes and viewing transformation to
* draw a line diagonally across the display.
                                                     HP 1000 AGP Supplement
```

I-9

\*

```
JSB J2MOV
                     ;absolute move
       DEF *+3
       DEF M1.0
                       x = -1.0
       DEF M1.0
                     ; y = -1.0
      JSB J2DRW
                     ;absolute draw
       DEF *+3
       DEF D1.0
                      x = 1.0
                     :
       DEF D1.0
                       y = 1.0
      JSB JWOFF
                     ;disable the work station
       DEF *+2
       DEF ID
                     ; work station 1
      JSB JWEND
                     ;terminate it
       DEF *+2
       DEF ID
                     ; work station 1
      JSB JEND
                     ;end the AGP system
       DEF *+1
      JSB EXEC
                     ;terminate this program
       DEF *+2
       DEF D6
      END USER
CDS Version:
MACRO,R,Q
      NAM USER
                    ;AGP User Program to draw a line
      CDS ON
                    ;enable code and data separation for use with
                    ;CDS versions of AGP libraries
      EXT JBEGN, JDINT, JWON, J2MOV, J2DRW, JWOFF, JWEND, JEND, EXEC
 ..Constants:
                    ;placed into data space
      RELOC DATA
      DEC 1
ID
                    ;work station id
WSPLEN DEC 8
                    ;work station name length
WSPNAM ASC 8, WSP1.RUN ; work station program name
                    ;graphics output LU
GLU
     DEC 1
CNTRL DEC 0
                    ;display initialization control word
M1.0 DEC -1.0
                    ;X and Y values of the beginning of the line
D1.0 DEC 1.0
                    ;X and Y values of the line's endpoint
D6
      DEC 6
                    ;exec code for program termination
 .. Main routine
                    ;placed into code space
      RELOC CODE
```

```
HP 1000 AGP Supplement
I-10
```

```
USER NOP
     PCAL JBEGN,0,0,0 ; initialize the graphics system
     PCAL JDINT, 5, 0, 0; initialize the work station
      DEF ID
                     ; work station 1
      DEF WSPLEN
                     ; work station name length=8
                     ; name = 'WSP1.RUN'
      DEF WSPNAM
      DEF GLU
                      ; LU of the graphics display device (LU = 1)
      DEF CNTRL
                     ; no control bits set
      PCAL JWON, 1, 0, 0; enable the work station for graphics output
       DEF ID
                     ; work station 1
 .. Everything between the asterisks may be changed as needed.
 .. Using default attributes and viewing transformation to
  draw a line diagonally across the display.
      PCAL J2MOV,2,0,0 ;absolute move
       DEF M1.0 ; x = -1.0
       DEF M1.0
                      ; y = -1.0
      PCAL J2DRW, 2,0.0 ;absolute draw
                       ; x = 1.0
       DEF D1.0
                       ; y = 1.0
       DEF D1.0
      PCAL JWOFF, 1, 0, 0 ; disable the work station
       DEF ID
                       ; work station 1
      PCAL JEND, 1, 0, 0 ; terminate it
                       ; work station 1
       DEF ID
      PCAL JEND,0,0,0 ;end the AGP system
      PCAL EXEC,1,0,0 ;terminate this program
       DEF D6
      END USER
```

## NAMING CONVENTIONS

#### Library File Names

This supplement refers to libraries by their filenames, or by the name they would be referred to if they were in the working directory.

Files containing device handlers in AGP are named so that the first character of the file name correlates with the logical function performed by the handler and the next four characters correlate with the physical device. For example, files beginning with "D" refer to a graphics display handler and files ending with "0019" refer to the HP 2623 Graphics Terminal. Specifically, the file named "D0019.LIB" contains the graphics display handler for the HP 2623A Graphics Terminal. For a complete list of device handlers supported by AGP and their corresponding numbers, refer to the *Device Handlers Manual*.

Table 2 shows the first character used for each logical device handler.

First Character	Logical Device
D	Display (Graphics)
А	Alphanumeric
L	Locator
Р	Pick
В	Button
ĸ	Keyboard
v	Valuator

Table 2. Device Handler File Naming Conventions

Internal User Routines begin with "J", "I", "Q", "M", and "F". Internal Common Blocks begin with "JO", "IO", "QO" and "MO". To avoid naming conflicts, do not name common blocks, subroutines or functions beginning with these letters.

The DGL device-independent library name is DIDD. The file containing this library is DIDD.LIB.

Libraries using code and data separation in the VC+ environment are suffixed by CDS.LIB. The CDS library for D0019.LIB would then be D0019\_CDS.LIB.

## LINKING AGP PROGRAMS

AGP is divided into two separate programs, with a monitor program (ZMNTR) running concurrently. Only one monitor program is needed to handle multiple AGP users.

#### PROGRAM

1) User Program (UP)

#### ADVANTAGES

Device independent. Can control any supported graphics configuration by initializing the appropriate WSP. Increased user code space.

- 2) Work Station Program (WSP) Configuration can be changed to include various peripherals without relinking the user program.
- 3) Monitor Program (ZMNTR) Cleans up the system resources and performs some WSP initialization tasks.

The User Program, Work Station Program and the Monitor are loaded separately and execute using class I/O to communicate. LINK is the loader used for RTE-6/VM and RTE-A. On the RTE-6/VM operating system, when the UP and ZMNTR are loaded, DECAR.LIB (the Decimal and String Arithmetic Library) must be searched. It is also recommended that link be sized as large as possible. This provides maximum space for relocation of all the external modules required to run AGP programs.

The following sections describe link procedures for UP, WSP, and ZMNTR and large user programs.

## The User Program

The UP is loaded by relocating the application program in the standard manner and searching the AGP User Library (UPLIB.LIB). The following link command sequence is typical:

RE, <program></program>	Relocate a	user program
SE, UPLIB. LIB	Search the	AGP User Program Library
EN	-	search the system library
	and end	

## The Work Station Program

Linking the WSP requires that some decisions be made first. The steps in

the process for a typical WSP are as follows:

- 1) Determine the WSP program name.
- 2) Determine the functions and graphics logical devices included, and
- 3) Link the WSP.

In addition, it is possible to configure the WSP for particular applications.

The following steps are required to link a WSP:

a) Relocate the WSP main program module, (i.e., WSP.REL or WSP\_CDS.REL for applications using CDS).

- b) Search the AGP WSP library (WSPLB.LIB). This step appends the SDA (Segment Display Area), their management routines and other AGP routines.
- c) Search one physical device handler file for each graphics logical function chosen (e.g., B0001.LIB for the HP 2623A Button). It is possible for one physical device to contain several logical graphics devices. The alpha, button, keyboard, locator, valuator and pick library may be searched in any order. However, the graphics display handler library must be the last library searched (e.g., D0019.LIB for the HP 2623A Graphics Display).
- d) Search the DGL library (DIDD.LIB). This appends all the DGL routines necessary to support the handlers. It also adds dummy handlers for any logical functions not explicitly searched in the previous step.
- e) Search the system library to pick up necessary system support routines.

In the following example, a non-CDS WSP is used. The WSP supports the graphics display, keyboard, and locator logical functions. The HP 2623 terminal is used for the graphics display and keyboard, and the HP 9111 tablet is the locator.

RE,WSP.REL	Relocate the standard WSP main program
SE,WSPLB.LIB	Search the WSP library
SE,KOOO1.LIB	Search the HP 2623 keyboard input handler file
SE,LOOO4.LIB	Search the HP 9111 locator input handler file
	Display handler must be searched last
SE,D0019.LIB	Search the HP 2623 graphics display handler file
SE, DIDD. LIB	Search the DGL library
EN	Implicitly search the operating system libraries
	and end

Note: If WSP\_CDS.REL is used, the CDS versions of graphics libraries must be searched.

## **EMA/VMA Device Handlers**

Some device handlers use EMA/VMA as indicated in the *Device Handlers* Manual. Refer to the following manuals for additional information on EMA/VMA:

RTE-A Programmer's Reference Manual RTE-6/VM Programmer's Reference Manual RTE-A Link Reference Manual RTE-6/VM Line Reference Manual

Two link examples are listed below.

The following is a sample link procedure for a WSP which accesses the HP 2932 printer with an EMA space of 400K words:

CI>LINK EM,400 EMA space of 400K words required by D0053 RE,WSP.REL Relocate the standard WSP main SE,WSPLB.LIB Search the WSP library SE,D0053.LIB Search the display handler for the HP 2932 printer SE,DIDD.LIB Search the DGL library EN

The following is a sample link procedure for a WSP which accesses the HP 2932 printer using VMA and an in-memory working set of 100K words:

CI>LINK VM VMA space WS.100 Working set size of 100K words RE,WSP.REL Relocate the standard WSP main SE,WSPLB.LIB Search the WSP library Relocate the application program RE, PROG1.REL Search the display handler for the HP 2932 printer SE,D0053.LIB SE, DIDD. LIB Search DGL library EN

Depending on the operating system, computer and handler library used, it is possible that the default WSP program is too large to link. If this happens, there are several options available, any combination of which can be used to decrease the WSP size:

- Use CDS versions of the WSP software.
- Use EMA instead of VMA since VMA includes more system code into the program than EMA.
- Reduce the size of the SDA.
- Reduce the size of user-modifiable tables (e.g., the hash table size).
- Remove unnecessary functionality (e.g., use the dummy SDA or dummy polygons).

Minimum options resulting in no loss of functionality are:

- Use EMA.
- Reduce the SDA size to 512 words.
- Reduce the hash table size to 102 words.

#### The Monitor Program

It is recommended that the monitor program be linked onto the system in order to prevent it from being removed by another user. The following link command sequence can be used to link the monitor program in RTE-A.

\* Specify a system utility
\*
SU
\*
\*
Relocate the file ZMNTL.REL (for RTE-A only)
\*
RE,ZMNTL.REL
\*
\* End the linker and output the ZMNTR on directory/PROGRAMS
\*
EN,/PROGRAMS/ZMNTR.RUN

For the RTE-6 operating systems, use ZMNTR.REL instead of ZMNTL.REL.

## Large User Programs (Non-CDS Environment)

Without CDS, some AGP user application programs are so large that the user program must be segmented. In segmenting AGP user programs, it is necessary that all common blocks be relocated with the main. COM.REL contains all the AGP common blocks. Relocating COM.REL makes AGP common blocks available to all segments so they can act on the same data.

The following is a typical link command sequence for a segmented user program with AGP graphics calls in the main program and in each segment.

RE, MAIN.REL Relocate the Main Program RE,COM.REL Relocate the Common Block Module Search the AGP User Library if SE, UPLIB. LIB ## there are AGP calls in the main SE Search system library RE, SEG1.REL Relocate first segment Search the AGP User Library if SE, UPLIB. LIB \*\* there are AGP calls in the segment SE Search system library # RE,SEG2.REL Relocate second segment Search the AGP User Library SE, UPLIB. LIB \*\* there are AGP calls in the segment Implicitly search the system library EN and end.

## SPOOLING

#### General

AGP allows the user to spool graphics commands, but does not perform the actual spooling operations. See the appropriate operating system manual for a discussion of spooling.

Spooling is enabled by setting bit 0 of the control word in the JDINT call. When bit 0 in the control word is not set, the initialization of the device includes verifying the range of the LU specified, making sure that the correct system driver is attached to the LU and finally making sure that the device requested is at the LU value (if possible). If spooling is enabled (bit 0 in control is set), AGP does not perform any of these checks. Note that when bit 0 is set, AGP is forced to be in immediate visibility mode.

By configuring the operating system to perform spooling and then enabling spooling in a AGP application program, an application program can insure that graphics commands will be sent to a graphics device or a disc file without interruption. Two applications can benefit from spooling graphics. First, when several users are attempting to send graphics commands to a device at the same time, this can be used to restrict the use of the device to one user at a time. With this mechanism the graphics device can be treated similar to a spooled line printer. Spooling can also be used to save graphics commands in a file. This file can then be sent to the graphics device at a later time without having program.

If the alphanumeric device and the graphics display device are identical, alphanumeric and graphics data can be spooled simultaneously; but it is necessary to enable the graphics display before enabling the alphanumeric display.

## **Device Output**

When several programmers are sending graphics commands to one device, for example a plotter, spooling can be used to lock the device so that only one user can access it at a time. In many non-graphics environments, spooling is commonly used in this manner to control the use of line printers. Spooling can only be performed to the device supported by the device handler linked with the AGP program. For example, if an HP 7470A device handler is used, the device at the spooled LU must be an HP 7470A plotter.

The steps required to run a AGP application program on RTE-A are as follows:

CI>SP,ON,GLU CI>PRGRAM,GLU CI>SP,OF,GLU

where GLU is the graphics LU

On RTE-6/VM, the sequence is:

:SL,GLU,,,GLU	*Set up GLU (graphics LU) to be spooled
:RU,PRGRAM,GLU	*Outspool graphics to GLU. The application
	<pre>* must enable spooling in JDINT</pre>
:CS,GLU	*Close spool file. The file will be sent
	* to the graphics device when it is
	* available for use.

## File Output

AGP output can be spooled to a disc file which can later be sent to the graphics/alphanumeric device. This is useful when a picture may need to be generated several times. It avoids having to rerun the application program each time to generate the picture.

To be used for spooling, an LU must be configured for spooling using the Batch Spool Monitor. RTE-A users should refer the Users Manual and RTE-6/VM users to the Batch Spool Monitor Manual for the specific steps required.

The steps required to run a AGP program on RTE-6/VM which spools to a file are as follows:

:CR,filename:SC:LU:3	*Create a file if one doesn't exist.
:SL,GLU,filename:sec:LU,BO	*Set up the file to be associated with
	* GLU on RTE-6, for both reading and writin
:RU, PRGRAM, GLU	*Run the AGP program. JDINT must
	<pre>* specify spooling.</pre>
:CS,GLU	*Close the spool file.

On RTE-A, the corresponding sequence is:

CI>SP,ON,GLU,filename CI>PRGRAM,GLU CI>SP,OF,GLU

Once created, the file can only be sent to a physical device which is the same type supported by the device handler linked with the application program. For example, if a file was created using an HP 7470A device handler, the file can only be sent to an HP 7470A plotter.

The spooled output is transferred from the file created to the destination device in a different manner for each destination device. Spooled files can only be sent to a graphics/alphanumeric display by using the FMGR "ST" command or the CI "CO" command together with a device-dependent binary code.

#### Using Graphics Terminals

To send the graphics output created by the device handler for a specific graphics terminal from a disc file, use one of the following:

CI>CO, namr, LU of the graphics terminal or FMGR:ST, namr, cntrl

where cntrl = 3000B + the LU of the graphics terminal. Note that cntrl cannot be an expression. CI should be used where available.

## Using HP Plotters or Translators

To send the graphics output created by the device handler for a specific HP Plotter from a disc file, use one of the following:

CI>CO,namr,LU of the plotter or FMGR:ST,namr,cntrl

where cntrl = 2100B + the LU of the HP Plotter. Note that cntrl cannot be an expression. CI should be used where available.

.

# Part II Advanced Topics

## **Description of AGP Programs**

#### USER PROGRAM

The user program (UP) is a program written by the user to specify the graphics task. It contains the following components:

- 1) The User Written Code is code written in the host language that contains calls to AGP subroutines that specify the graphics task.
- 2) The AGP Command Modules generate virtual coordinate data from the user-specified world coordinate data. Only those routines referenced by user-specified AGP calls and their support routines are loaded into the user program.
- 3) The AGP Communication Module handles communications between the AGP modules in the user program and those in the WSP.

The UP can communicate with up to eight WSPs simultaneously in AGP. Each WSP drives its own set of devices.

#### WORK STATION PROGRAM

The work station program (WSP) is a program that receives commands and virtual coordinate data from the AGP command modules and performs the graphics tasks specified in the commands. It can be thought of as a logical graphics device that performs graphics tasks as requested by the user program.

The work station program:

- 1) maintains the necessary data structures to perform all actions that may be requested of it by the user program.
- 2) contains a set of device handlers that perform all I/O functions to a specified set of physical graphics devices.

The devices supported by a fully equipped WSP are:

- 1) Alphanumeric display
- 2) Button
- 3) Graphics display
- 4) Keyboard
- 5) Locator
- 6) Pick
- 7) Valuator

The WSP contains the following components:

- 1) The WSP communication and control module communicates with the AGP modules in the user program and calls the appropriate AGP modules in the WSP to process all requests from the user program.
- 2) The segment display area is a data structure used to store information about graphics segments (e.g., the primitives comprising a segment).
- 3) The segment display area manager consists of all AGP code used to process and inquire about the segment display area (e.g., the code used to perform software picking and to redraw the picture.
- 4) The *device handler code* consists of one device handler for each logical AGP device.

#### MONITOR PROGRAM

The Monitor (ZMNTR) performs several actions as part of the WSP initialization process and also as a monitor for cleaning-up system resources. The program performs the following functions:

- 1. Allocates all the class numbers that AGP uses.
- 2. "Clones" a WSP from a type 6 file or a permanently loaded program and schedules it as part of the WSP initialization process.
- Removes the ID segment of the WSP(s) it has RP'd after it terminates on the RTE-6/VM system. (On RTE-A, this action is performed by the operating system.)

4. Cleans up class resources after a user program or work station program terminates abnormally. (On RTE-A, this action is performed by the operating system.)

## Abnormal Program Termination

In most cases, if an AGP user program is terminated without a call to JEND (i.e. OF'ed) all the work station programs associated with that user program are terminated.

The segment display area extension, if one exists, may remain open. Any font files which were open at the time will remain  $\phi$ pen.

If a work station program is abnormally terminated (i.e. OF'ed), the segment display area extension remains open, if one exists. AGP will generate an error of level 4 "Graphics aborted".

On RTE-6/VM, ZMNTR assumes a program has been terminated if it no longer exists, or it is in state 0 (dormant) and is not on the time list.

On RTE-A, the operating system determines when a program has been terminated and performs the clean-up. If, however, the user program is terminated while a work station program is waiting on a read from a device, the operating system does not clean up the work station program.

#### CONCURRENT WSP SCHEDULING

Frequently, two or more user programs may wish to use the same WSP concurrently or a user program may wish to use the same WSP for multiple work stations. AGP accomplishes this by cloning copies of the WSP as needed. Cloning is the process of making a copy of a program with a different name. AGP does the cloning of the WSP transparently to the user so the user need not worry about making copies of the same WSP. To allow AGP to make a clone, one of the following procedures must be used:

1) Load the WSP as a permanent program (RTE-6/VM only), or

2) Save the WSP as a type 6 file.

Assuming one of these procedures is used, AGP can make renamed copies of the WSP available to all AGP user programs. AGP has a renaming convention for the WSP. AGP will try 12 times as follows to WSP that is not already in use:

1) AGP will try to make the last two characters the true LU from which the program was run (i.e "WSPXX" will be transformed to "WSPLU"

where LU is the last two digits of the logical unit that is running the program). If the work station program name is less than three letters, periods will be used to fill in the blank spaces (i.e. "AB" will change to "AB.LU" and "A" will change to "A..LU").

- 2) If the name from step one is already in use, the next try is to put a period in the third character of the WSP name if this was not done previously in step one.
- 3) Finally, if there is still a naming conflict, AGP will try changing the third character to the following values in order until an unused name is found: "/", "0", "1", "2", "3", "4", "5", "6", "7", "8", "9". Since exactly 12 tries will be made, the "9" will only be tried in the case of a one, two or three character name where step 2 above was skipped.

For example, if the work station program name is "WSPXX" and the LU is 12, the following names will be tried: "WSP12", "WS.12", "WS/12", "WS012", "WS112", ... "WS812". If the work station program name is "A" and the LU is 15 the following names will be tried: "A..15", "A./15", "A.015", "A.115", ... "A.915".

Problems may occur due to the automatic cloning of WSPs by AGP. For example, if the program is scheduled from LU 48, the WSP could end up with the name "WSP48" and could be confused for another program with a type 6 file of the same name.

## AGP Program-to-Program Communications

The user program and the work station program must be able to communicate with each other when they are running concurrently. In RTE this program-to-program communication is done through class I/O. AGP's use of class I/O is transparent to the user since the subroutines handle the allocation of class numbers and all class I/O communication.

AGP ensures that all system resources it uses are properly returned to the system upon program termination. Normally, the user program and the WSP return all the resources they use as part of their termination process. If a user program or WSP is terminated abnormally, the monitor program on RTE-6/VM or the operating system on RTE-A performs the class I/O clean-up.

#### CLASS I/O ALLOCATION

Two class numbers are allocated each time a WSP is initialized. The values of these class numbers are passed to the user program and the WSP. The user program uses one class number to transmit all messages to the WSP, and the

WSP uses the other class number to transmit messages to the user program. Normally, these class numbers are returned when the user written code requests that the WSP be terminated through a JWEND call.

## The Segment Display Area

The Segment Display Area (SDA) is the AGP data structure used for maintaining segments. Whenever a new segment is created (JOPEN), a segment entry is made in the SDA. As output primitives and primitive attribute changes are incorporated into the segment, a virtual coordinate representation of these primitives and attributes are recorded in the SDA. Closing a segment (JCLOS) closes the entry in the SDA.

Along with output primitives and attributes, the SDA entry maintains information on the segment attributes. This information is updated as the application program changes the segment attributes.

Once a segment is closed, it cannot be reopened. Primitives may not be added to or changed in the closed segment. The only operations that can be made to a closed segment are segment deletion, or changing the dynamic segment attributes of highlighting, visiblility, and detectablity.

Each work station has its own SDA space. The SDA space for a work station is located in memory and can be extended to a disc file if necessary, using the JSDF call.

A segment remains in the SDA until:

- 1) It is explicitly deleted (JPURG)
- 2) All segments are deleted (JCLR)
- 3) The last work station on which it appears is terminated (JWEND)
- 4) The currently open segment overflows and therefore becomes too large to remain in the SDA.

Output primitives which are created within a segment are stored in the SDA. Output primitives created outside of a segment will not be stored in the SDA except in a batch-of-updates (see JBATC).

For applications using little or no segments, the reservation and allocation of SDA file space will not normally be of concern to the AGP application programmer. However, if the application programmer requires too many segments or any very large segments, the program may request additional space be reserved for the SDA by using the JSDF call to open a disc file (see JSDF). The programmer will also need to reserve additional space if a large number of primitives are issued outside of segments while in batchof-updates (see JBATC).

#### SEGMENT DISPLAY AREA OVERFLOW

During the construction of a segment, a segment display area overflow may occur. The overflow error will not be reported until the segment is closed (JCLOS). To inquire if the open segment has overflowed on a work station, the inquiry JIWS (Opcode 1002) may be used. This test will only work while the segment is open.

Before JCLOS, which actually closes the overflowed segment, any calls which affect that segment (JRNAM, JSVIS, JSDET, JSHI, JVSAL) will generate errors and will be ignored on that work station. None of the mentioned calls will cause a new-frame-action on the overflowed work station.

When a segment overflows, the segment is removed from the SDA of the work station on which it overflowed. This is done without generating a newframe-action. The remaining image on the display becomes primitives outside of a segment.

After a segment overflows and is still open, it is treated similarly to primitives outside of a segment on the work station where the overflow occurred. The primitives may still retain characteristics of the segment, (e.g. if the segment was invisible, the primitives may be invisible, if the segment was highlighted, the primitives may be highlighted). At the first occurrence of a new-frame-action the primitives that were specified as part of the overflowed segment will be removed from the display as if they were primitives outside of a segment. An overflowed segment is not pickable. It should be noted that attempting to change an attribute of an overflowed segment will not generate a new-frame-action since the segment is not treated as a segment in the data base.

The current attributes assigned to the open segment will be applied to any primitives added to a segment once it has overflowed. However, the primitives will not be added to the SDA.

An SDA overflow can also occur during a batch-of-updates. When this happens there is no room left in the SDA for new batched data and the "batched" output is sent to the graphics display device. Subsequent data is then sent to the graphics display device as if there were no batching in effect on that work station. All other work stations will remain unaffected.

#### CHANGING THE SIZE

To increase performance when using segments or batch-of-updates the user may increase the size of the "in-memory" segment display area at the load time of a WSP. Also at load time the user has control over the maximum size that the Segment Display Area may be (which may include a segment display area disc extension). To control these sizes, the user must edit

and compile block data subroutines shipped with AGP and then relocate the modified version before searching the WSP library (WSPLB).

The block data subroutine KOSDF (shipped as source in KOSDF.FTN) holds the data area for the SDA buffer. This data area is used by the WSP to hold part of the SDA. The default value (the value obtained when not modified by the user) is 2048 words which is 16 FMGR blocks (128 words each).

When KOSDF.FTN is modified, the following rules must be followed:

- 1) Only the dimension of the buffer (SDFBUF) and the DATA statement for the word containing this buffer dimension (SDFDIM) should be changed. If they are changed, they must both be changed so SDFDIM has the same value as the dimension of the buffer (SDFBUF).
- 2) SDFBUF's dimension (and SDFDIM's value) must be a multiple of 512 words which is 4 FMGR blocks of 128 words each.

There are no checks to verify that user editing was done correctly. If the rules above are not followed, unpredictable results may occur.

A WSP's maximum possible SDA size (including the disc extension) is determined by the block subroutine KOPAG (shipped as source in KOPAG.FTN). KOPAG is used as an array of pointers and flags which contain information about the SDA. For each 512 words (4 FMGR blocks) in the SDA, 3 words of information are stored in KOPAG. The default size tained when not modified by the user) is 510 words. This allows the maximum SDA to be (510/3)\*4 or 680 blocks of 128 words KOPAG.FTN.

When modifing KOPAG.FTN the following rules must be followed:

- 1) Only the dimension of the buffer (BUFFER) and the DATA statement for the word containing this buffer's dimension (BUFDIM) should be changed. If they are changed, they must both be changed so BUFDIM has the same value as the dimension of the buffer (BUFFER).
- 2) BUFFER's dimension must be a multiple of 3. For each 3 words of BUFFER, the SDA maximum size is increased by 4 blocks.
- 3) The maximum SDA size must be less then 8188 blocks so BUFFER size must be less than 6141 words.
- 4) The maximum SDA size set by KOPAG must not be less than the "in memory" SDA buffer size set in KOSDF, that is:

BUFDIM\*512. ----->= SDFDIM. 3

It is recommended that the "in memory" SDA size (as determined by KOSDF) be set as large as possible if increased performance is desired. The maximum

size possible for KOSDF depends on the size that a WSP can load in. Thus, it is recommended that CDS versions of the relocatables and libraries be used on VC+ operating systems.

The following is a sample CDS linker file sequence for relocating the new modules when changing the segment display area size in the VC+ environment:

RE,WSP\_CDS.REL Relocate the WSP main program RE,KOSDF\_CDS.REL Relocate the new SDA buffer size (compiled w/\$CDS on) RE,KOPAG\_CDS.REL Relocate the new maximum SDA size (compiled w/\$CDS on) SE,WSPLB\_CDS.LIB Search the WSP library (CDS version) SE,DIDD\_CDS.LIB Search the HP 2623 Graphics Display handler (CDS version) SE,DIDD\_CDS.LIB Search the DGL library (CDS version) EN Implicitly search the system library and end

### LINKING A WSP WITHOUT A SEGMENT DISPLAY AREA

A module called SDUM.REL (SDUM\_CDS.REL for CDS use in VC+ operating systems) can be relocated during the loading of the WSP which will satisfy the requirements of the WSP and provide a dummy segment display area manager.

This is very useful when writing interactive programs which do not use segments and need to run in a smaller partition because of memory limitations.

If the user program attempts to open a segment, or start a batch-ofupdates, an error will occur telling the user that segment operations are not possible in the WSP.

The following is a typical linker command sequence when linking the WSP without a segment display area manager.

RE,WSP.REL	Relocate the WSP program
RE, SDUM. REL	Relocate the Dummy SDA module
SE,WSPLB.LIB	Search the WSP library
SE,D0019.LIB	Search the HP 2623 graphics display handler file
SE, DIDD. LIB	Search the DGL library
EN	Implicitly search the system library and end

### The Segment Hashing Table

AGP uses hashing to access segments by name (JOPEN, JRNAM, JPURG, JSVIS, etc.). The segment name is hashed to an entry in a "segment hashing table" (or "hashing table"), and from there AGP searches a linked list of segments in the SDA.

### **CHANGING THE SIZE**



The user can control the performance of AGP segment hashing by altering the size of the segment hashing table. It is the nature of the hashing algorithm that for a given number of segments a larger hashing table usually means shorter linked lists and quicker access by name.

Changing the hashing table size is done via a mechanism similar to that provided for changing the SDA size. That is, the user must edit a special FORTRAN common block which defines the hashing table, changing its dimensions to the desired size. The user must then compile the common block, and relocate it into a new WSP.

The hashing table common block is KONTB (shipped as source in KONTB.TFN). It contains a variable, NTABSZ, and an array NTABLE. NTABLE provides the actual space for the hashing table, and NTABSZ gives AGP a way of knowing the size of NTABLE. The user should edit KONTB.FTN and change both the DATA statement for NTABSZ and the DIMENSION statement for NTABLE. Currently the default for NTABSZ is 202.

AGP requires two words (i.e., two spaces in NTABLE) for each entry in its hashing table. Therefore, the dimensioned size of NTABLE and NTABSZ must be an even number. The number of entries in the AGP hashing table will be NTABSZ/2. Hashing runs most efficiently with no more than one segment per hashing table entry, and efficiency decreases as more segments are created for each hashing table entry.

The user should abide by the following rules in changing KONTB:

- 1) NTABSZ must be an even number, greater than zero; if it is not, the work station will abort at run-time with error 37.
- 2) NTABSZ must be equal to the dimensioned size of NTABLE. Errors are not reported if it is not equal, and results in this case are unpredictable.
- 3) For fastest hashing, a) NTABSZ should be about twice the expected total number of segment names and b) NTABSZ should be twice a prime number. While neither is necessary, each contributes to efficiency.
- 4) If the user expects to create a large number of segments, it may also be necessary to change the size of the SDA.

For example, if NTABSZ=202 and NTABLE(202) were specified, a hashing table with 101 entries would be provided. NTABSZ is twice a prime number (101) and would be most efficient if no more than 101 segments were created. Increasing the number of segments to more than 101 would decrease efficiency somewhat, as would sizing NTABSZ at twice a non-prime number, such as 200, instead of 202.

The following is a typical link command sequence for linking a new KONTB:

RE,WSP.REL	Relocate the WSP program
RE,KONTB.REL	Relocate the new KONTB
SE,WSPLB.LIB	Search the WSP library
SE,D0019.LIB	Search the HP 2623 graphics display handler file
SE,DIDD.LIB	Search the DGL library
EN	Implicitly search the system library and end

### Polygon Set Efficiency Considerations

There are four ways to modify polygon set functionality in order to enhance the efficiency of AGP:

- o Change the size of the polygon style table.
- o Change the size of the intercept buffer.
- o Change the size of the vertex buffer.
- o Suspend polygon set functionality (to reduce the WSP size).

#### CHANGING THE SIZE OF THE POLYGON STYLE TABLE

The default polygon style table for each device contains 16 entries defining 16 different styles. The size of the table can be changed to accomodate more (or fewer) than 16 styles. To change the size of the polygon style table, the user must edit and compile the source block data subroutine Z1PTB.FTN which is shipped with DGL, and relocate the modified version before searching the DGL libraries at WSP load time.

The example below shows the link file sequence for relocating the module:

RE,WSP.REL	Relocate the WSP main program
SE,WSPLB.LIB	Search the WSP library
RE,Z1PTB.REL	Relocate the polygon style table
SE,D0019.LIB	Search the 2623 Graphics Display handler
SE,DIDD.LIB	Search the DGL library
EN	Implicitly search the system library and end

The maximum number of indices should not exceed 255, and the minimum must be at least 1.

Upon initialization of the WSP, the default table is initialized. Any new entry beyond the sixteenth is defined to represent the same polygon style as index 1. These entries can then be re-defined with JDPST.

If the table contains less than 16 entries, only those indices in the table are initialized; that is, the table is initialized as a strict, ordered subset of the default table.

NOTE: This table should never be sized to zero.

### CHANGING THE SIZE OF THE INTERCEPT BUFFER

As each interior fill line is computed, the intersections of the fill line with the polygon set boundary are stored in the DGL part of the WSP; in other words, intersection coordinates are stored in an intercept buffer. The default maximum number of intersections which can be stored for each fill line is 100. If a fill line produces more than this maximum number of intersections, it cannot be drawn. Other lines within the same polygon set can still be drawn, however, as long as the number of their intersections are within range. The size of the intercept buffer can be changed to accommodate a wider range of intersections per fill line. It should never be reduced to hold the coordinates of less than two intersections.

If the application program uses only concave polygons, and thus generates fill lines which intersect the boundary only twice, the buffer could be reduced to hold only two intersections per fill line. This would save a substantial amount of program space for the WSP.

If, on the other hand, the application program uses extremely complex polygon sets, it might be necessary to enlarge the buffer. (The maximum number of boundary intersections for any fill line never exceeds the number of vertices of the clipped polygon set.)

To change the size of the intercept buffer, the user must edit and recompile the source block data subroutine T1INT.FTN shipped with DGL and relocate the modified version before the DGL libraries are searched at WSP load time. The example below shows the link file sequence for relocating the module:

RE,WSP.REL	Relocate the WSP main program
SE,WSPLB.LIB	Search the WSP library
RE,T1INT.REL	Relocate the intercept buffer
SE,D0019.LIB	Search the 2623 Graphics Display handler
SE, DIDD. LIB	Search the DGL library
EN	Implicitly search the system library and end

#### CHANGING THE SIZE OF THE VERTEX BUFFER

The vertices of a clipped polygon set are stored in the vertex buffer of the WSP. The default WSP vertex buffer holds up to 100 vertices.

If, after clipping, more vertices are generated than a work station's vertex buffer can hold, Error 57 is generated, and the polygon set is not filled on that work station.

The size of the vertex buffer can be changed to accommodate a wide range of vertices. It should never be reduced to hold less than one vertex.

To change the size of the vertex buffer, the user must edit and recompile the source block data subroutine K1FIL.FTN shipped with AGP and relocate the modified version before the WSP library is searched at WSP load time. The example below shows the link file sequence for relocating the module:

RE,WSP.REL	Relocate the WSP main program
RE,K1FIL.REL	Relocate the WSP vertex buffer
SE,WSPLB.LIB	Search the WSP library
SE, D0019.LIB	Search the 2623 Graphics Display handler
SE, DIDD. LIB	Search the DGL library
EN	Implicitly search the system library and end

#### LOADING DUMMY POLYGON SET MODULES

The user can partially or totally suspend polygon set functionality (that is, replace the polygon set modules with dummy modules) to reduce the amount of program code space used in the WSP.

PARTIAL SUSPENSION AT THE DGL LEVEL. For AGP users, suspension of polygon set functionality at the DGL level suspends the filling capability only. An outline of the boundary of the clipped polygon set is generated.

To suspend polygon set functionality at this level the user must relocate the source block data subroutine PGNDM.REL (PGNDM\_CDS.REL for CDS use in VC+ operating systems) shipped with DGL before the DGL libraries are searched at WSP link time. The following example shows the loader file sequence for relocating the module:

RE,WSP.REL	Relocate the WSP main program
RE, PGNDM. REL	Relocate the polygon set module
SE,WSPLB.LIB	Search the WSP library
SE,D0019.LIB	Search the 2623 Graphics Display handler
SE, DIDD. LIB	Search the DGL library
EN	Implicitly search the system library and end

Polygon set calls are affected by the suspension as follows:

CALLS	EFFECT
J2PGN, J3PGN, JR1PG, JR3PG	Error 57 is reported. An outline of the boundary of the clipped polygon is generated.
JDPST	Error 60 is reported. The call is ignored.
JPICL, JPILS	Error 60 is reported. The call is ignored.

The following information is returned for the inquiry call JIWS:

Opcode 1065	No polygon calls supported.
Opcode 1066	Default polygon style.
Opcode 1067	Default polygon interior color. in hardware.
Opcode 1069	No immediate retroactive change of polygon style.
Opcode 1070	No hardware generation of polygons.

FULL SUSPENSION AT THE WSP LEVEL. For full suspension, polygon set functionality must be suspended at the WSP level. To suspend polygon set functionality at this level the user must relocate the source block data subroutine WPGDM.REL (WPGDM\_CDS.REL for CDS use in VC+ operating systems) before searching the WSP library at WSP load time.

The following example shows the link file sequence for relocating the module.

Full suspension means no polygon set functionality will be available. This should only be used when polygons are not going to be used by the user program.

RE,WSP.RELRelocate the WSP main programRE,WPGDM.RELRelocate the polygon set dummy modulesSE,WSPLB.LIBSearch the WSP librarySE,D0019.LIBSearch the 2623 Graphics Display handlerSE,DIDD.LIBSearch the DGL libraryENImplicitly search the system library and end

Polygon set calls are affected by the suspension as follows:

CALLS	EFFECT	
J2PGN, J3PGN, JR2PG, JR3PG	Error 57 is reported. is generated.	No polygon
JDPST	Error 60 is reported. is ignored.	The call

The following information is returned for the inquiry call JI1IN:

Opcode	20	Normal results.
Opcode	21	Normal results.
Opcode	22	Normal results.

The following information is returned for the inquiry call JIWS:

Opcode 1065 No polygon calls supported.

Opcode 1066

Opcode 1067

Default polygon interior color.

Default polygon style.

Opcode 1068 No polygon points supported in hardware.

Opcode 1069

No immediate retroactive change of polygon style.

Opcode 1070

No hardware generation of polygons.

### Color Modelling Efficiency Considerations

#### CHANGING THE SIZE OF THE COLOR TABLE

Any graphics output device with a color table has a default color table, which is loaded whenever the corresponding device handler is loaded in an AGP program. The default tables of specific devices are given in the Device Handlers Manual.

To change the size of the color table, the user must edit and recompile the source block data subroutine Z1CTB.FTN and relocate the modified version before the DGL libraries are searched at WSP load time. The example below shows the link file sequence for relocating the module:

RE,WSP.REL	Relocate the WSP main program
SE,WSPLB.LIB	Search the WSP library
RE,Z1CTB.REL	Relocate the relocatable file
SE,D0019.LIB	Search the 2623 Graphics Display handler
SE, DIDD. LIB	Search the DGL library
EN	Implicitly search the system library and end

For most devices with color tables, the number of colors in the default table can be increased or decreased at load time. This number should be greater than zero and less than 32767 (the system limit). (Each table includes color 0, the background color.) Note that for devices that allow redefinition of the background color the number must be greater than one.

If the number of entries in the color table is increased, then each new entry has the same default parameters as entry 1. For example, if the default table has 10 entries, and its size is increased by five, then entries 11-15 have the same color definition as entry 1. Therefore, the

effect of referencing entry 12, for instance, will be the same with the modified table as with the default table. To redefine the contents of new entries, use JDCOL.

If the size of the default table is decreased by n entries, then the last n entries are deleted.

#### LOADING DUMMY COLOR MODELLING ROUTINES

If the color modelling capabilities of AGP are not needed, dummy color modelling routines can be loaded; this frees most of the memory space in which the real routines would otherwise be loaded and stored.

To link dummy color modelling routines, the user must relocate the source block data subroutine COLDM.REL (COLDM\_CDS.REL for use in VC+ operating systems) shipped with DGL before the DGL libraries are searched at WSP load time. The following example shows the link file sequence for relocating the module.

RE,WSP.REL	Relocate the WSP main program
SE,WSPLB.LIB	Search the WSP library
RE, COLDM. REL	Relocate the color modelling module
SE, D0019.LIB	Search the 2623 Graphics Display handler
SE, DIDD. LIB	Search the DGL library
EN	Implicitly search the system library and end

If this is done, all color modelling capabilities are gone, including the ability to redefine color tables; but the ability to choose colors from the (default) color table of the graphics output device, using JCOLR or JPICL, remains. The calls that assume color modelling, JCOLM, JDCOL and JICOL are ignored. A call to JIWS (Opcode 1075) returns a zero. This does not mean that the color table has no entries, but that none of them can be changed; this gives the program a way to find out whether color modelling capabilities are available.

Calls to JIWS, with the opcodes specific to color modelling, return appropriate information:

Opcode 1071	No retroactive color change.
Opcode 1072	No change of background color.
Opcode 1073	No color change.
Opcode 1074	Default color model: RGB.
Opcode 1075	Color modelling unavailable.

### Fonts

The following is a listing of the file names representing the font style for each of the six high quality text fonts (stored on disc) which are supplied with AGP:

File Name	Font Style
FONT1	Eurostyle
FONT2	Simplex Roman
FONT3	Triplex Roman
FONT4	Script
FONT5	Mathematical
FONT6	Gothic

Refer to the AGP Reference Manual for a detailed description of each font style.

.

· · · · ·

.

.

·

.

.

# Part III System-Dependent Calls

### INTRODUCTION

The calls described in this supplement contain system-dependent information. These descriptions, together with the (system-independent) call descriptions in the AGP Reference Manual provide complete information for AGP subroutines.

The format of the following pages is the same as the format in the Reference Manual. You may insert these call descriptions in the AGP Reference Manual (in alphabetical order) or use the system-specific manual as is.

JDFNT

### JDFNT

PURPOSE: To associate a font number with a font file for use with high quality text. CALLING SEQUENCE: CALL JDFNT (FONT, SLANT, FNLEN, FNAME, CONTRL) [ INTEGER; Input ] FONT FONT number to be associated with the FONT file specified. Used in the JFONT call to select a desired font. ( 1 <= FONT <= 16) SLANT [ REAL; Input ] Angle in radians specifying the character slant. Range is from -PI/2.0 to +PI/2.0 with a value of 0.0 specifying straight up.  $(-PI/2.0 \le SLANT \le PI/2.0).$ FNLEN [ INTEGER; Input ] The number of characters from FNAME to be used as the name of the font file. FNAME [ INTEGER; Array; Input ] The name of the file containing the desired font, in Packed ASCII format. [ INTEGER; Input ] CONTRL Used to control the characteristics of the character production or font file. Bits should be set according to the following bit map. All unused bits are reser-

ved for future use and should be set to zero.

### JDFNT

- BIT 0 = 0 AGP will leave the disc font file open as much as possible, closing it only when necessary. At most, one font file will be open at any time. This is the most efficient method of accessing the font file, since the number of disc accesses (for font file opens and closes) is limited.
  - = 1 AGP will close the font file after every call which accesses it. This mode of operation is much more inefficient and uses more system resources than would otherwise be needed. This mode the possibility of of operation will limit the possibility of open upon abnormal program.

BIT 1-15 - Currently unused. Should be set to 0.

JDFNT associates FONT with the specified slant and filename. If FONT was previously associated with a font file, JDFNT will associate the new font file specified with the specified font number.

Whenever the font number is referenced for subsequent high quality text output, the character data in the specified font file will be used. Note that a font must be set to the current font (JFONT) and defined (JDFNT) before it can be used for high quality text.

The current font has a disc font file associated with it via the JDFNT call. Typically this disc file is left open by AGP as much as possible to increase performance, with the restriction that at most one font file, (the one associated with the current font) is open at any one time. If it is important that the font file not be left open there are two ways of assuring it is closed. If BIT 0 is set, the particular disc font file will be closed after every AGP routine that accesses it. Alternatively, if BIT 0 is not set any open font file can be closed at any time via the JFONT call (see JFONT for details). The FONT files are not opened exclusively and therefore may be accessed by concurrent user programs.

SLANT is used to generate slanted text. The major use of this parameter is to generate italicized text. The angle specified must be between -PI/2.0 and +PI/2.0. A value of 0.0 for SLANT will generate characters that are not slanted.

For more information on high quality text font files, see Part I of this supplement.

# JDFNT

ERROR CONDITIONS:

064 Level 0: AGP was not initialized by JBEGN.

INFO: Undefined. ACTION: An implicit call to JBEGN is made.

123 Level 3: Incorrect number of parameters specified for the call.

INFO: Number of parameters passed. ACTION: Call ignored.

090 Level 3: The font number specified is out of range. Possibly a non-positive number or real number was passed.

INFO: The font requested. ACTION: Call ignored.

125 Level 3: The character count in FNLEN is less than one, or greater than 63. Possibly a negative or real number was passed.

INFO: Character count specified. ACTION: Call ignored.

089 Level 2: The slant is out of range. The slant must be in the range -PI/2.0 to +PI/2.0.

INFO: Undefined. ACTION: A value of 0.0 is used.

-2XXX Level 3: FMP error number -XXX. The FMP error is due to a problem with the font file.

INFO: Undefined. ACTION: Call ignored.

088 Level 3: Specified file is not a font file.

INFO:	Undefined.
ACTION:	Call ignored.

## JDINT

### JDINT

PURPOSE: To initialize a work station.

CALLING SEQUENCE: CALL JDINT(ID, WSPLEN, WSPNAM, LU, CONTRL)

ID [INTEGER; Input] The ID used to refer to this work station in subsequent AGP calls. (1 <= ID <= 8.)

WSPLEN [ INTEGER; Input ]

The number of characters from WSPNAM to be used as the name of the work station program.

WSPNAM [ INTEGER; Array; Input ] The name of the work station program to be scheduled, in Packed ASCII format.

> [ INTEGER; Input ] The logical unit to be associated with the graphics display of the work station. If a graphics display is not explicitly loaded, this parameter is ignored. The LU cannot be mapped to a file with FORTRAN file I/O commands.

CONTRL

LU

[ INTEGER; Input ] Used to control the characteristics of the work station. Bits should be set according to the following bit map. All unused bits should be set to 0.

| 0| 0| 0| 0| 0| 0| 0| 0| 0| 0| 0| 0| X| X| 0| X| 15 14 13 12 11 10 9 8 7 6 5 4 3 2 1 0

BIT 0 = 0 - No system spooling will be possible to the LU. = 1 - LU may be outspooled using system spooling.

> As part of the work station initialization process, AGP verifies (if possible) that the device at the LU is of the proper type for the work station program. To do this it sends an inquiry to which the device responds with its particular identification sequence. Setting bit 0 inhibits the inquiry process. AGP does

# JDINT

not perform spooling. A LU must be configured to be outspooled using system spooling routines. Input will not be possible when outspooling is in use. Setting this bit causes all subsequent attempts to enable an input device at LU to be flagged with errors and ignored. Not all graphics displays can be spooled. Refer to the *Device Handlers Manual* to determine if a particular graphics display device can be spooled.

- BIT 1 = 0 Currently unused. Set to 0.
- BIT 2 = 0 LU will not be locked by AGP.

= 1 - LU will be locked by AGP. Setting Bit 2 will lock the LU of the graphics display device. This will prevent any I/O, not from this work station, and some operating system prompts, from reaching the LU.

When this feature is used some operating system commands may not function correctly, since they may not be able to send their I/O to the LU. Locking the LU uses some system resources. See the appropriate operating system manual for additional information on LU locking.

The LU is not locked by the AGP system until after the graphics output device is initialized. If two programs try to initialize a device simultaneously, unpredictable results may occur even though one or both may set bit 2.

#### BIT 3 = 0 - Simulated raster erase will not be enabled.

= 1 - Simulated raster erase will be enabled. Some graphics output devices have the ability to erase images. AGP will use this feature when the simulated raster erase bit is set in the JDINT call. This feature allows AGP to do such things as remove purged segments without clearing the graphics display device, and redraw all visible segments. Note: Erasing is done by redrawing the image in the background color. Therefore, lines crossing the image to be purged may also have parts of them erased thus leaving "holes". See the Device Handlers Manual for the effect of bit 3 on a particular device.

BITS 4-6 - Currently unused. Set to 0.

- BIT 7 = 0 Immediate erase. AGP will clear the graphics display during initialization.
  - = 1 No erase. Whatever was on the display stays there. This action is device-dependent. See the Device Handlers Manual for the effect of bit 7 on a particular device.
- BITS 8-15 Currently unused. Set to 0.

JDINT initializes a work station. A work station must be initialized before the application program can reference it for any purpose. During this call an attempt will be made to ensure that the device at the LU specified is really the type of display device that is loaded into the specified WSP. In some cases it is impossible to determine whether or not the device is the correct type (e.g., the HP 1350 Graphics Translator). Therefore, the errors that refer to whether the device is the correct one may not be reported. Furthermore, when the spooling bit is set, no checks are made as to the validity of the LU.

Unpredictable results may occur if there is no device at the LU, if the device is not the intended graphics device, or if the device is turned off or in a local mode.

If the graphics output device is a dummy device handler the control word CONTRL will be ignored.

If the graphics device is not outspooled, the device name is set to the name of the device as specified in the *Device* Handlers Manual. If the graphics output device has the outspooling bit set, then the name is set to the last four characters of the display library followed by two blanks. For example, the name for the HP 2623 graphics display device handler would be "0001 ".

AGP cannot verify that the program specified is a work station program. If it is not, AGP errors will occur when the user program attempts to communicate with it.

No other work station or program should send any I/O to the LU while it is initialized by this work station. There is no AGP error check for the above condition and unpredictable results may occur if it is violated.

Note that AGP makes an implicit call to JMCUR as part of this call.

Before an application program can use one of AGP's logical devices, it must first be enabled. Each logical device can be enabled and disabled independently of the others. A call to JWON enables graphics output, and a call to JWOFF disables graphics output. Similarly, a call to JEDEV selectively enables one of AGP's other logical devices, and a call to JDDEV disables it. A user program may initialize up to eight work stations at any one time. For some operating systems there is also a limit to the number of active work station programs on the system at one time (see Part III of this supplement for more information regarding system resources.)

If the Work Station Program (WSP) has been loaded permanently or saved as a type 6 file, AGP schedules a copy of the WSP. In this way, one or more user programs can use copies of the same WSP.

The following example demonstrates initialization and termination of a work station. It also shows how one enables an input device, in this case the locator.

CALL JDINT(1,9,9HW2623.RUN,LU,0) CALL JWON(1)	*Initialize the work station *Enable graphical output
:	mante Brahmear eacher
CALL JEDEV(1,4,LOCLU)	*Initialize the locator
{do locator input}	
CALL JDDEV(1,4)	*Terminate the locator
CALL JWOFF(1)	*Disable graphical output
CALL JWEND(1)	*Terminate the work station

ERROR CONDITIONS:

064 Level 0: AGP was not initialized by JBEGN.

INFO: Undefined. ACTION: An implicit call to JBEGN is made.

123 Level 3: Incorrect number of parameters specified for the call.

INFO: Number of parameters passed. ACTION: Call ignored.

074 Level 3: Work station ID out of range.

INFO: ID of the work station. ACTION: Call ignored.

127 Level 3: The character count in WSPLEN is less than one or greater than 63. Possibly a negative or real numer was passed.

INFO: Character count specified. ACTION: Call ignored.

## JDINT

096 Level 3: The ID number specified has been used previously for another work station.

INFO: ID number specified. ACTION: Call ignored.

009 Level 2: This routine should not be called while a segment is open. Possibly a call to JCLOS was omitted.

INFO: Name of the open segment. ACTION: An implicit call is made to JCLOS.

003 Level 3: This routine should not be called during a batch of updates. Possibly missing a JUPDT.

> INFO: Undefined. ACTION: An implicit call is made to JUPDT.

116 Level 4: The monitor program ZMNTR could not be found. ZMNTR must have an ID segment to be scheduled.

INFO: ID of the work station. ACTION: Graphics aborted.

153 Level 4: The monitor program ZMNTR terminated abnormally.

INFO: ID of the work station. ACTION: Graphics aborted.

154 Level 4: Incorrect ZMNTR for this system. Reload the proper ZMNTR for the operating system.

INFO: Undefined. ACTION: Graphics aborted.

109 Level 3: Illegal WSP name.

INFO: ID of the work station. ACTION: Call ignored.

-3XXX Level 3: FMP error -XXX. The FMP error occurred while trying to access the work station program specified.

INFO: ID of the specified work station. ACTION: Call ignored.

### JDINT

119 Level 3: Checksum error in WSP type 6 file (probably the type 6 program was loaded on another system), or the file is not a type 6.

INFO: ID of the work station. ACTION: Call ignored.

114 Level 3: No more ID segments available to clone the specified WSP.

INFO: ID of the work station. ACTION: Call ignored.

113 Level 3: All the available cloning names for the WSP are already in use.

INFO: ID of the work station. ACTION: Call ignored.

115 Level 3: Specified WSP program not found. It has no ID segment and no type 6 file.

> INFO: ID of the work station. ACTION: Call ignored.

117 Level 1: No copy bit set in WSPs ID segment. The specified WSP cannot be cloned. Possibly the WSP program is currently executing.

INFO: ID of the work station. ACTION: Call ignored.

079 Level 3: No class number is available.

INFO: ID of the work station. ACTION: Call ignored.

093 Level 3: The maximum number of work stations for a given CPU have already been initialized. Some other work station must be terminated through a call to JWEND before the specified work station can be initialized.

> INFO: ID of the work station. ACTION: Call ignored.

092 Level 3: The specified work station program cannot be scheduled.

INFO: ID of the work station. ACTION: Call ignored.

149 Level 3: Revision code mismatch between the UP and the WSP (e.g., the UP was loaded with new software and the WSP was loaded with old software).

> INFO: ID of the work station. ACTION: Call ignored.

037 Level 4: NTABSZ, in common block KONTB of the work station program, is not an even, positive number. Possibly the user has incorrectly modified the common block. (NTABSZ defines the number of entries in the hashtable for access of segments by name.)

> INFO: ID of the specific work station. ACTION: AGP aborted.

094 Level 3: The LU specified does not match the device driver in the WSP, the device at the LU is down, the LU is not in the user's session LU table, or the EQT associated with the LU is down. Possibly the wrong driver was loaded upon creation of the WSP or the incorrect LU was specified.

> INFO: LU specified. ACTION: Call ignored.

135 Level 3: There are no resource numbers available to lock the output device.

> INFO: The output LU. ACTION: Call ignored.

136 Level 3: The attempt to lock the output LU has failed.

INFO: The output LU. ACTION: Call ignored.

148 Level 3: The requested work station program is not an AGP work station program.

INFO: ID of the work station. ACTION: Call ignored.

### JEDEV

PURPOSE: To enable a logical device (other than the graphics display).

CALLING SEQUENCE: CALL JEDEV (ID, CLASS, LU)

ID [ INTEGER; Input ]
The ID of the work station whose logical device is to
be enabled.
(1 <= ID <= 8)</pre>

CLASS [ INTEGER; Input ] The class of the logical device which is being enabled:

> CLASS = 1 - Alphanumeric device = 2 - Button device = 3 - Keyboard device = 4 - Locator device = 5 - Valuator device = 6 - Pick device

LU [ INTEGER; Input ] The logical unit of the device. The LU must represent a physical device. The LU cannot be mapped to a file with FORTRAN file I/O commands.

JEDEV enables a logical device on a particular work station. The logical device should be enabled before it is used (e.g., the pick device must be enabled (JEDEV) before it can be used for input (JPICK)). A logical device is terminated through a call to JDDEV. The work station on which the logical device is being enabled, should be initialized (see JDINT). A logical device is enabled and disabled independently of the graphics output being enabled and disabled (see JWON, JWOFF).

When enabling the locator or pick device, the locator or pick echo reference position is set to its default value (see JLOCP or JPIKP).

No logical input device will be enabled if the LU is the same logical unit as a graphics display device which is being outspooled (see JDINT).

If the specified locator/pick is not the same LU as that of an initialized graphics display device, then the logical locator/pick limits will be set to the default values for the particular locator used. If the LU specified is the same as that of an initialized graphics display device, then the logical locator/pick limits are set to the current view surface limits.

Not all logical devices are available on all work stations. The actual logical devices that are available on a particular work station are determined at the time the Work Station Program (WSP) is loaded. JIWS can be used to inquire if a logical device is supported by a particular work station. The LU and the device will also be verified by AGP and the name of the device will be set accordingly. This information is also available to the user program through the JIWS call.

No other work station or program should send any I/O to the LU while it is initialized by this work station. There is no AGP error check for the above condition and unpredictable results may occur if it is violated.

Example:

CALL JDINT(3,9,9HW2623.RUN,LU,0) CALL JWON(3)	<pre>#initialize work station #3 #enable graphical output</pre>
CALL JWOFF(3)	*disable graphical output
CALL JEDEV(3,4,LOCLU)	*enable the locator
CALL JWLOC(3,1, BUTTN, VX, VY)	*do input
CALL JWON(3)	*enable graphical ouput
CALL JWLOC(3,1,BUTTN,VX,VY)	*do input
CALL JDDEV(3,4)	*terminate the locator
CALL JWOFF(3) CALL JWEND(3)	*disable graphical output *terminate work station

ERROR CONDITIONS:

064 Level 0: AGP was not initialized by JBEGN.

INFO: Undefined. ACTION: An implicit call to JBEGN is made.

123 Level 3: Incorrect number of parameters specified for the call.

INFO: Number of parameters passed. ACTION: Call ignored.

074 Level 3: Work station ID out of range.

INFO: ID of the work station. ACTION: Call ignored.

## JEDEV

012 Level 3: The logical class is out of range.

INFO: Logical class requested. ACTION: Call ignored.

071 Level 3: The work station specified has not been initialized by JDINT. Possibly there is a call to JDINT missing, or the ID is not identical to that used in the JDINT call.

> INFO: ID of the work station. ACTION: Call ignored.

003 Level 3: This routine should not be called during a batch of updates. Possibly missing a JUPDT.

> INFO: Undefined. ACTION: An implicit call is made to JUPDT.

121 Level 2: Attempt to enable a logical device that is already enabled at ID.

> INFO: ID of the work station. ACTION: Previously enabled logical device is disabled with an implicit call to JDDEV.

005 Level 3: The work station specified has a dummy device handler loaded for that logical device. An inquiry (see JIWS) of the work station may be made to determine its capabilities.

> INFO: ID of the work station. ACTION: Call ignored for that work station.

080 Level 3: Attempt to enable an input function on an outspooled LU. Input can only be performed from an LU rather than that of an outspooled graphical display.

> INFO: ID of the work station. ACTION: Call ignored.

094 Level 3: The LU specified does not match the device driver in the WSP, the device at the LU is down, the LU is not in the user's session LU table, or the EQT associated with the LU is down. Possibly the wrong driver was loaded upon creation of the WSP or the incorrect LU was specified.

INFO: LU specified. ACTION: Call ignored.

### JIWS

PURPOSE: To inquire	about some characteristic of a specific work station.			
CALLING SEQUENCE:	CALL JIWS(ID, OPCODE, ISIZE, RSIZE, <u>ILIST</u> , <u>RLIST</u> )			
ID	[ INTEGER; Input ] The ID of the work station to which the inquiry is to be directed. (1 <= ID <= 8)			
OPCODE	[ INTEGER; Input] The code specifying which characteristic of the work station is to be queried. The supported values of OPCODE are listed below.			
ISIZE	[ INTEGER; Input ] The number of integer parameters to be returned in ILIST by the inquiry function.			
RSIZE	[ INTEGER; Input ] The number of real parameters to be returned in RLIST by the inquiry function.			
ILIST	[ INTEGER; Array; Output ] An array of INTEGER characteristics of the work sta- tion returned as a function of OPCODE. All strings are returned in Packed ASCII format.			
RLIST	[ REAL; Array; Output ] An array of REAL characteristics of the work station returned as a function of OPCODE.			
JIWS allows the application programmer to inquire several characteristics about the specified work station. All characters returned are in Packed ASCII format.				

The thousands digit of the OPCODE parameter specifies the number of integer values returned in ILIST; the hundreds digit specifies the number of real values returned in RLIST. It is the responsibility of the application program to insure that ILIST is dimensioned at least as large as specified by ISIZE, and that RLIST is dimensioned at least as large as specified by RSIZE. If ISIZE or RSIZE is not the exact size required by a particular inquiry function then the call is ignored and an error is reported. An exception to these rules is found in OPCODE 15001, where a variable length SDF extension file pathname is returned in ILIST. For this opcode, the fifth integer returned is really a pathname. Therefore, ILIST should be

dimensioned to be 36 elements long (4 integers plus up to a 63 character pathname, packed 2 characters per word). The ILIST value passed for this opcode is 5.

JIWS implicitly makes the picture current before making the inquiry. JIWS may only be called if the work station is initialized.

The following codes are available:

OPCODE DESCRIPTION

201 Virtual coordinates of last PICK point (Returns (0.,0.) if no PICK has been performed)

RLIST(1) = X virtual coordinate RLIST(2) = Y virtual coordinate

247 Resolution of pick device

RLIST(1) = Resolution in X direction (points/mm) RLIST(2) = Resolution in Y direction (points/mm)

248 Maximum dimensions of pick device

RLIST(1) = Maximum size in X direction (mm) RLIST(2) = Maximum size in Y direction (mm)

249 Current pick echo position

RLIST(1) = X virtual coordinate RLIST(2) = Y virtual coordinate

- 250 For internal use only
- 251 Marker size in virtual coordinates.

RLIST(1) = Marker width in virtual coordinates RLIST(2) = Marker height in virtual coordinates

252 Resolution of graphics display.

RLIST(1) = Resolution X direction (points/mm)
RLIST(2) = Resolution Y direction (points/mm)

253 Maximum dimensions of graphics display

RLIST(1) = Maximum size in X direction (mm) RLIST(2) = Maximum size in Y direction (mm)

## JIWS

254	Aspect ratios	
	RLIST(1) = Aspect ratio of the virtual coordinate system RLIST(2) = Aspect ratio of the logical displ	217
	limits	ay
255	Resolution of locator device	
	RLIST(1) = Resolution in X direction (points/mm) RLIST(2) = Resolution in Y direction (points/mm)	
256	Maximum dimensions of locator device	
	RLIST(1) = Maximum size in X direction (mm) RLIST(2) = Maximum size in Y direction (mm)	
257	Current locator echo position	
	RLIST(1) = X virtual coordinate RLIST(2) = Y virtual coordinate	
258	For internal use only.	
259	For internal use only.	
450	For internal use only.	
451	For internal use only.	
1001	Does the work station have an SDA?	
	ILIST(1) = 0 - No, the dummy SDA was loaded. ILIST(1) = 1 - Yes	
1002	Has the open segment overflowed?	
	ILIST(1) = -1 no open segment ILIST(1) = 0 SDA has not overflowed ILIST(1) = 1 the open segment has overflowed	
	Can the graphics device do simulated raster erase?	
~	ILIST(1) = -1 - No ILIST(1) = 0 - Yes, but it is not enabled ILIST(1) = 1 - Yes, and it is enabled	
,		

# JIWS

· · ·

1050	Does the graphics display device support hardware clipping at physical limits?
	<pre>ILIST(1) = 0 - No ILIST(1) = 1 - Yes, at the view surface boundaries. 2 - Yes, but only to the physical limits of the device.</pre>
1051	Justification of view surface within the logical display limits.
	<pre>ILIST(1) = 0 - View surface is centered within logical display limits. ILIST(1) = 1 - View surface is positioned in the lower- left corner of the logical display limits.</pre>
1052	For internal use only
1053	Number of distinct colors supported on the graphics display.
	ILIST(1) = Number of distinct colors supported.
1054	Maximum number of distinct colors that can appear on graphics dis- play at one time.
	<pre>ILIST(1) = Number of distinct colors that can appear on display device at one time.</pre>
1055	For internal use only.
1056	Linestyles supported by the graphics display
	ILIST(1) = Number of hardware linestyles supported
1057	Linewidths supported by the graphics display.
	ILIST(1) = Number of linewidths supported.
1058	Hardware character sizes supported by the graphics display.
	ILIST(1) = Number of character sizes supported. ILIST(1) = -1 continuous character sizes.
1059	Number of markers supported by graphics display.
	ILIST(1) = Number of distinct markers supported.
1060	For internal use only.
1061	For internal use only.
1062	For internal use only.
1063	For internal use only.
HP 1000 A III-18	AGP Supplement

1064 For internal use only.

1065 Number of polygon styles supported on the graphics display.

- ILIST(1) > 0 Number of polygon styles.
  - = 0 No polygon manipulation capabilities available (i.e., polygon routines were suspended at the DGL level).
    - =-1 No polygon manipulation capabilities available (i.e., polygon routines were suspended at the WSP level).
- 1066 For internal use only.
- 1067 For internal use only.
- 1068 Number of polygon vertices supported by the display device.

ILIST(1) = Number of vertices. ILIST(1) = 0 - no hardware support. ILIST(1) = MAXINT - no hardware limit on number of vertices.

NOTE: MAXINT is the maximum integer size (CPU dependent).

1069 Indicates whether or not the display device supports immediate retroactive change of polygon style.

ILIST(1) = 0 - No = 1 - Yes

1070 Indicates whether or not the display device supports hardware generation of polygons.

1071 Indicates whether or not the device supports immediate retroactive color change.

ILIST(1) = 0 - No 1 - Yes

1072 Indicates whether or not background color can be redefined.

ILIST(1) = 0 - No= 1 - Yes

# JIWS

1073	Indicates whether or not the device supports modification to color table.			
	ILIST(1) = 0 - No $= 1 - Yes$			
1074	Indicate which color model is in use.			
	ILIST(1) = 1 - RGB $= 2 - HSL$			
1075	Indicates size of color table.			
	ILIST(1) = Size			
1076	Current polygon interior linestyle.			
	ILIST(1) = Current polygon interior linestyle index.			
2050	For internal use only.			
3001	WSP name			
	ILIST(1)-ILIST(3) = Six character WSP name as restored (RP'd).			
5050	Graphics display device information			
	<pre>ILIST(1)-ILIST(3) = Six character string containing device name ILIST(4) = status</pre>			
	ILIST(5) = LU of graphics display device = 0 if a dummy display device is loaded			
5051	Keyboard device information			
	<pre>ILIST(1)-ILIST(3) = Six character string containing device name. ILIST(4) = status</pre>			
	ILIST(5) = LU of keyboard device = 0 if keyboard device is not enabled			

s. Na ze

HP 1000 AGP Supplement III-20

Ľ

6049	Pick device information
	<pre>ILIST(1)-ILIST(3) = Size character string containing device name. ILIST(4) = status</pre>
	<pre>= 0 disabled = 1 enabled. ILIST(5) = LU of pick device = 0 if no pick device is loaded = 0 if no pick device is loaded</pre>
6050	Locator device information
	ILIST(1)-ILIST(3) = Six character string containing the device name.
	<pre>ILIST(4) = status = -1 if there is a dummy locator device loaded = 0 disabled.</pre>
	= 1 enabled. ILIST(5) = LU of locator device
	= 0 if no locator device is enabled ILIST(6) = Number of associated buttons
	= 0 if no locator device handler is loaded
6051	Button device information
	<pre>ILIST(1)-ILIST(3) = Six character string containing device name ILIST(4) = status</pre>
	= 1 enabled ILIST(5) = LU of button device
	= 0 if not button device enabled
	ILIST(6) = Number of buttons on the button device
	= 0 if no button device handler is loaded
7050	Alphanumeric device information
	<pre>ILIST(1)-ILIST(3) = Six character string containing device name.</pre>
	ILIST(4) = status = -1 if there is a dummy alphanumeric device loaded = 0 disabled
	= 1 enabled ILIST(5) = LU of alphanumeric device
	= 0 if no alphanumeric device enabled ILIST(6) = maximum number of lines displayable
	<pre>= -1 if undeterminable (i.e. line printer) ILIST(7) = maximum number of characters per line</pre>

.

# JIWS

7051	Valuato	or device	information
	ILIST(	1)-ILIST(3	) = Six character string containing device name.
	ILIST()	4) = statu: = -1 if = 0 di: = 1 en:	s there is a dummy valuator device loaded sabled
	ILIST(	•••	valuator device no valuator device enabled
	ILIST(	6) = Numbe:	r of associated buttons
	ILIST(	7) = Numbe	no valuator device handler loaded r of subvaluators
		= 0 if :	no valuator device handler loaded
15001	Segment d	isplay area	a and extension
	ILIST(2)	= Usable m	e size in blocks (128 words) emory resident SDA size in blocks (128 word) f unused blocks (128 word) (in memory and/or on
	disc) ILIST(4) = Number of characters in the disc file name. A zero is returned if there is no extension file.		
	ILIST(5)		r string containing the disc file name.
ERROR	CONDITIONS:		
	064 Level 0:	AGP was n	ot initialized by JBEGN.
		INFO: ACTION:	Undefined. An implicit call to JBEGN is made.
	123 Level 3:	Incorrect	number of parameters specified for the call.
		INFO: ACTION:	Number of parameters passed. Call ignored.
	023 Level 1:	ID out of	range.
		INFO: ACTION:	ID of the work station. Call ignored.
	070 Level 1:	Possibly	station specified has not been initialized. a call to JDINT is missing, or the ID is not to that used in the JDINT call.
		INFO: ACTION:	ID of the work station. Call ignored.

L

į.

**HP 1000 AGP** Supplement III-22

· ·· · r

JIWS

055 Level 1: The requested opcode is not available. Possibly negative or real number was passed.

> INFO: Opcode requested. ACTION: Call ignored.

054 Level 1: ISIZE is not equal to the number of integer parameters to be returned. Possibly a negative or real number was passed.

> INFO: ISIZE given. ACTION: Call ignored.

059 Level 1: RSIZE is not equal to the number of real parameters to be returned. Possibly a negative or real number was passed.

> INFO: RSIZE given. ACTION: Call ignored.



**JSDF** 

JSDF

PURPOSE: To create a disc file to be used as an extension to the segment display area.

CALLING SEQUENCE: CALL JSDF(ID, NAMLEN, NAME, SIZE, CONTRL)

ID [INTEGER; Input] The ID of the work station to which this call applies.

NAMLEN [ INTEGER; Input ] The number of characters from NAME to be used as the name of the file.

NAME [ INTEGER; Array, Input ] The name of the file used for the extension, in Packed ASCII format.

SIZE [ INTEGER; Input ] The size in blocks (128 words each) of the file that is to be created. If size = 0, the AGP system default of 100 blocks will be used.

CONTRL [ INTEGER; Input ] Used to control characteristics of the segment display file extension. Bits should be set according to the following bit map. All unused bits should be set to zero.

> | 0| 0| 0| 0| 0| 0| 0| 0| 0| 0| 0| 0| X| 0| 0| 0| 15 14 13 12 11 10 9 8 7 6 5 4 3 2 1 0

BIT 0 - 2 - Currently unused.

BIT 3 = 0 - Give error if disc file specified exists. = 1 - Purge and recreate the file if it exists.

BIT 4 -15 - Currently unused.

JSDF creates a (FMGR type 1) disc file that AGP uses as an extension to the segment display area. If the file already exists and bit 3 of the CONTRL parameter is set, AGP purges it and recreates the file with the file size specified.

JSDF may only be called while the specifed work station is initialized. JSDF may be called only once for each work station.

If AGP uses up all the memory space it has in the segment display area and no extension is available, all further actions requiring SDA storage generate errors until the situation is remedied.

The file size given in the JSDF call should be a multiple of 4. AGP will reduce the size given to a multiple of 4 if a non-multiple is given. No errors will be given for this condition.

The file area that is defined with the JSDF call should be larger than the "in memory" SDA size by at least 4 FMGR blocks. If the requested file size is smaller than that in core it will not be used and an error will be generated. The "in memory" SDA size can be inquired through a call to JIWS with an opcode of 15001. The default "in memory" SDA size (used unless a special module is relocated with the WSP) is 2048 words or 16 blocks.

The file size given in the JSDF call cannot be larger than the maximum allowed by that work station. If the requested size is larger than the maximum, the file size will be set to the maximum and an error will be given. The default maximum SDA disc file size (used unless a special module is relocated with the WSP) is 680 FMGR blocks. For more information regarding the segment display area see Part I of this supplement.

A remedy for an SDA overflow may be to call JSDF if it has not been previously called for that work station.

The SDA extension file is closed and purged by the AGP system, when the work station is terminated (JWEND).

ERROR CONDITIONS:

064 Level 0: AGP was not initialized by JBEGN.

INFO: Undefined. ACTION: An implicit call to JBEGN is made.

123 Level 3: Incorrect number of parameters specified for the call.

INFO: Number of parameters passed. ACTION: Call ignored.

074 Level 3: Work station ID out of range.

INFO: ID of the work station. ACTION: Call ignored.

# **JSDF**

071 Level 3: The work station specified has not been initialized by JDINT. Possibly there is a call to JDINT missing, or the ID is not identical to that used in the JDINT call.

> INFO: ID of the work station. ACTION: Call ignored.

129 Level 3: The character count in NAMLEN is less than one or greater than 63. Possibly a negative or real number was passed.

INFO: Character count specified. ACTION: Call ignored.

026 Level 3: The SDF size given is less than zero. Only non-negative integer numbers should be passed.

INFO: The size given. ACTION: Call ignored.

035 Level 3: An enabled work station does not support segment or batch-of-update operations.

INFO: The ID of the work station which does not support the needed operation. ACTION: Call ignored.

091 Level 3: JSDF can only be called once for each work station. Once the segment display area extension is set up, its size cannot be changed.

> INFO: ID of the work station. ACTION: Call ignored.

156 Level 3: The file size given in the JSDF call is smaller than this work station can use. The file must be at least 4 blocks larger than the portion of the SDF that is in memory.

> INFO: Minimum SDF size that could be used by the work station. ACTION: Call ignored.

-4XXX Level 3: FMP error number -XXX. The FMP error is due to a problem with the segment display area.

INFO: ID of the work station. ACTION: Call ignored.

HP 1000 AGP Supplement III-26

- 157 Level 2: The file size given in the JSDF call is larger than this work station can use.
  - INFO: Maximum SDF disc file size that could be used by the work station.
  - ACTION: The SDF disc file size is set to the maximum that the work station can use.

**JSERR** 

## JSERR

PURPOSE: To set the error handling conditions.

CALLING SEQUENCE: CALL JSERR (RLU, RLEVEL, ALEVEL) RLU [ INTEGER; Input ] The LU number to which error output is sent (i.e. the error log LU). RLEVEL [ INTEGER; Input ] The minimum severity error level that will be reported.  $(0 \leq \text{RLEVEL} \leq 4)$ ALEVEL [ INTEGER; Input ] The minimum severity error level that results in "aborting" AGP.  $(0 \leq ALEVEL \leq 4).$ 

JSERR sets the error handling conditions of reporting and AGP termination and implicitly makes the picture current.

JSERR allows the application program to determine which errors will be reported to the "error log LU" (RLU). All errors of severity level greater than or equal to the error report level (RLEVEL) will be reported. Initially the error report level equals 0, (i.e., all errors will be logged on the RLU). If the error report level is set to 4, only level 4 errors will be logged. If this feature is used before a program is completely debugged, important information may be missed because the error messages will not be printed.

At AGP initialization the RLU is set to the LU from which the program was scheduled.

Setting ALEVEL allows the application program to recover from certain AGP detected errors that do not exceed the specified severity level. Errors greater than or equal to ALEVEL will terminate graphics. ALEVEL = 0 implies all errors and warnings are fatal; ALEVEL = 4 implies no recoverable errors are fatal. At AGP initialization ALEVEL is set to 4.

After AGP has aborted graphics, due to an error equal to or exceeding ALEVEL, all AGP calls will be ignored except for JBEGN, JIERR, and JEND.

The last AGP error can be inquired at any time by the JIERR call.

HP 1000 AGP Supplement III-28 JSERR is useful when the application program is using programmatic error recovery.

The severity level and recovery action (if any) for under the "Error Conditions" subheading of each Appendix A of the AGP Reference Manual lists all levels, the subroutines where they are reported, and the recovery performed when that error occurs. The RLU is specified as a valid LU number. It cannot be mapped to a file with FORTRAN file I/O commands.

The RLU is not checked to verify that it is a valid LU. If an error occurs that should be reported, AGP attempts to report it on the RLU. If this fails, AGP attempts to report the error message on the LU from which the program was run (the default error LU). If that attempt fails the error is not reported; however, the error can still be inquired via a call to JIERR.

ERROR CONDITIONS:

064 Level 0: AGP was not initialized by JBEGN.

INFO: Undefined. ACTION: An implicit call to JBEGN is made.

123 Level 3: Incorrect number of parameters specified for the call.

INFO: Number of parameters passed. ACTION: Call ignored.

043 Level 1: The report level is out of range. Possibly a real or negative number was passed.

INFO: Level requested. ACTION: Call ignored.

044 Level 1: The abort level is out of range. Possibly a real or negative number was passed.

INFO: Level requested. ACTION: Call ignored.

. .

# Part IV AGP Installation

# SYSTEM PREREQUISITES FOR INSTALLATION

There are certain system requirements for installing the AGP programs onto your system. Information regarding these requirements are listed below:

• ID Segment One ID segment available for each User Program, Work Station Program and Monitor program.

• Class Numbers All operating systems require two class numbers for a UP/WSP pair. The system will hang when AGP attempts to perform I/O functions if class numbers are not available.

(System Available Memory) 256 words required by the user program:

6 words - header information 11\*n where n is the number of WSPs

Example: (with 2 WSPs)

6 + 2 \* 11=28

ZMNTR - One class number is required to store information with a class read/write.

An error will be generated if more than 22 active WSPs are on one CPU at a time.

UPLIB may be generated into the system to avoid the need to specify it directly in the load process.

You must not generate the device handler files D0001, WSPLB and DIDD into the system.

Most efficiency is gained from separate memory partitions for the UP, ZMNTR and WSP. Beneficial when the system disc is a mini-floppy or floppy disc.

• User Libraries

• SAM

• Memory Partitions

HP 1000 AGP Supplement IV-1

## INSTALLATION PROCEDURE

Follow the steps listed below to install AGP, Version 2.0, from the distribution media onto your system.

- Insert the first microfloppy disc, mag tape, or CS/80 cartridge tape into the tape drive. Determine the LU (or CRN for microfloppies) of the drive. This is referred to as the <source LU>. Mount the cartridge for microfloppies.
  - Note: The tape media you have received is in the TF (Tape Filer) format. You may wish to consult your HP 1000 System Utilities Manual for more detailed instructions on TF usage.
- 2. For mag tapes and CS/80 cartridges only:

Enter the Command Interpreter and run TF. The following TF command replaces any graphics software you presently have under the <destination mask> from your media and verifies that the software was copied.

CI> TF TF: co <source LU> <destination mask> dv

Examples:

Graphics software is copied into the current working directory from the mag tape drive at LU 8.

TF: co 8 @.@ dv

Graphics software is copied into the /GRAPHICS directory from the CS/80 cartridge tape drive at LU 24.

TF: co 24 /graphics/@.@ dv

3. For microfloppies only:

The software on microfloppies is shipped under the /GRAPHICSV2 directory. After entering CI and mounting the volume for the microfloppy, enter the following command to extract the graphics software from the microfloppy to the <destination mask> you specify. This command replaces any graphics software you presently have at your <destination mask> and verifies the transfer.

ci>:co/graphicsv2/@.@ <destination mask> d

Dismount the microfloppy volume, remove the microfloppy, insert the next microfloppy and repeat the process until all the microfloppies have been copied.

Examples:

HP 1000 AGP Supplement IV-2

Graphics software is copied from the microfloppy to the present working directory.

CI> co,/graphicsv2/@.@,@.@,dv

Graphics software is copied from the microfloppy to the directory /graphics.

CI> co,/graphicsv2/@.@,/graphics/@.@,dv

4. For all media:

The software numbering file: A92862, shipped with the graphics product, contains a list of all of the product's modules and a short one line description of each. Since the graphics product includes software for both CDS and non-CDS operating systems, this software number file is useful if you want to be selective about the modules you put on your disc.

The convention used in the graphics products is an \_CDS suffix for compiled modules for use in the VC+ operating system environment. Sources and font files shipped with the product can be used in either environment.

> HP 1000 AGP Supplement IV-3

# Appendix A AGP Subroutine Changes

## Summary of AGP Subroutine Changes

Changes in the format of the graphics calls from AGP Version 1.0 to AGP Version 2.0 are due to changes in the file system. The table below catagorizes the calls that are impacted.

Category	AGP Call	Reason for Change
Initialization	JDINT	Work station program names may exceed 5 characters.
Font File Selection	JDFNT	Font file names may exceed 6 characters. The security and LU parameters no longer exist with the hierarchial file system.
Segment Display Area Extension	JSDF	The disc file extension to the segment display area exceed 6 characters.
Inquiry	JIWS	Opcode 8001 is replaced by opcode 15001.

# **Specific Changes**

Presented below are the calls and parameter changes that accompany the AGP port from the non-hierarchial HP 1000 file systems (e.g., RTE-IVB, RTE-A.1) to the hierarchial file systems on RTE-A and RTE-6/VM (revision C.83 and later). For a comprehensive discussion of the routines, refer to the AGP Reference Manual and the AGP Version 2.0 Supplement for HP 1000 Systems.

## 1. JDINT

Version 1.0 (ID, WSPNAM, LU, CONTRL)

Version 2.0 (ID, WSPLEN, WSPNAM, LU, CONTRL)

HP 1000 AGP Supplement A-1 Parameter additions: WSPLEN=character length of WSP name

Parameter changes: WSPNAM=wsp name name extensions are not assumed.

## 2. JDFNT

Version 1.0 (FONT, SLANT, FNAME, SEC, CRN, CONTRL)

Version 2.0 (FONT, SLANT, FNLEN, FNAME, CONTRL)

Parameter additions: FNLEN=font file name length

Parameter deletions: SEC=security code of font file CRN=cartridge reference LU

Parameter changes: FNAME=font file name - name extension are not assumed

## 3. JSDF

Version 1.0 (ID, NAME, SIZE, CONTRL)

Version 2.0 (ID, NAMLN, NAME, SIZE, CONTRL)

Parameter additions: NAMLN=file name length

Parameter changes: NAME=segment display area file name - name extensions are not assumed

## 4. JIWS

Version 1.0 (ID, OPCODE, ISIZE, RSIZE, ILIST, RLIST)

Version 2.0 OPCODE changed

Parameter changes: OPCODE of 15001 returns the name of the segment display extension area file name. OPCODE 8001 should not be used.

# READER COMMENT SHEET

Advanced Graphics Package Version 2.0 Supplement for HP 1000 Systems

92862-90001 May 1984

We welcome your evaluation of this manual. Your comments and suggestions help us to improve our publications. Please use additional pages if necessary.

Is this manual technically accurate?		Yes	No No
Are the concepts and wording easy to understand?		Yes	No No
Is the format of this manual convenient in size, arrangen readability?	nent and	Yes	No No
Comments:			
		. •	
		Date:	
FROM:			
Name			
Company			
Address			

LD	FOL
	NO POSTAGE NECESSARY IF MAILED IN THE UNITED STATES
BUSINESS REPLY MAIL FIRST CLASS PERMIT NO. 1070 CUPERTINO, CALIFORNIA	

FOLD

FOLD

## Product Line Sales/Support Key

## Key Product Line

- A Analytical CM Components
- C Computer Systems Sales only
- CH Computer Systems Hardware Sales and Services
- CS Computer Systems Software Sales and Services
- E Electronic Instruments & Messurement Systems
- M Medical Products
- MP Medical Products Primary SRO
- MS Medical Products Secondary SRO
- P Personal Computation Products
- Sales only for specific product line
- " Support only for specific product line

tMPORTANT: These symbols designate general product line capability. They do not insure sales or support availability for all products within a line, at all locations. Contact your local sales office for information regarding locations where HP support is available for specific products.

HP distributors are printed in italics.

## **HEADQUARTERS OFFICES**

If there is no sales office listed for your area, contact one of these headquarters offices.

### NORTH/CENTRAL AFRICA

Hewlett-Packard S.A. 7, Rue du Bois-du-Lan CH-1217 MEYRIN 2, Switzerland Tel: (022) 83 12 12 Telex: 27835 hpse Cable: HEWPACKSA Geneve

#### ASIA

Hewlett-Packard Asia Ltd. 6th Floor, Sun Hung Kai Centre 30 Harbour Rd. G.P.O. Box 795 HONG KONG Tel: 5-832 3211 After Jan. 1, 1984 47th Floor, China Resources Bldg. 26 Harbour Rd., Wanchai HONG KONG Telex: 66678 HEWPA HX Cable: HEWPACK HONG KONG

CANADA Hewlett-Packard (Canada) Ltd. 6877 Goreway Drive MISSISSAUGA, Ontario L4V 1M8

Tel: (4 16) 678-9430 Telex: 6 10-492-4246 EASTERN EUROPE

Hewlett-Packard Ges.m.b.h. Liebigasse 1 P.O.Box 72 A-1222 VIENNA, Austria Tel: (222) 2365110 Telex: 1 3 4425 HEPA A

NORTHERN EUROPE Hewlett-Packard S.A. Uilenstede 475 P.O.Box 999 NL-1180 AZ AMSTELVEEN The Netherlands Tel: 20 437771

#### SOUTH EAST EUROPE Hewlett-Packard S.A.

7, Rue du Bois-du-Lan CH-12 17 MEYRIN 2, Switzerland Tel: (022) 83 12 12 Telex: 27835 hpse Cable: HEWPACKSA Geneve

#### **OTHER EUROPE**

Hewlett-Packard S.A. P.O. Box 150, Rie du Nant-D'Avril CH-1217 MEYRIN 2, Switzerland Tel: (022) 83 8111 Telex: 22486 hpsa Cable: HEWPACKSA Geneve

## MEDITERRANEAN AND

MIDDLE EAST Hewlett-Packard S.A. Mediterranean and Middle East Operations Atrina Centre 32 Kifissias Ave. Paradissos-Amarousion, ATHENS Greece Tel: 682 88 11 Telex: 21-6588 HPAT GR Cable: HEWPACKSA Athens

#### EASTERN USA Hewlett-Packard Co. 4 Choke Cherry Road

4 Choke Cherry Hoad ROCKVILLE, MD 20850 Tel: (301) 258-2000

## MIDWESTERN USA

Hewlett-Packard Co. 5201 Tollview Drive ROLLING MEADOWS, IL 60008 Tel: (312) 255-9800

**SOUTHERN USA** Hewlett-Packard Co. 2000 South Park Place P.O. Box 105005 **ATLANTA**, GA 30348 Tei: (404) 955-1500

#### **WESTERN USA**

Hewleti-Packard Co. 3939 Lankershim Blvd. P.O. Box 3919 LOS ANGELES, CA 91604 Tel: (213) 506-3700 OTHER INTERNATIONAL AREAS

#### Hewlett-Packard Co. Intercontinental Headquarters 3495 Deer Creek Road PALO ALTO, CA 94304 Tel: (415) 857-1501 Telex: 034-8300 Cable: HEWPACK

# SALES & SUPPORT OFFICES

Arranged alphabetically by country

#### ANGOLA

Telectra Empresa Técnica de Equipamentos R. Barbosa Rodrigues, 41-1 DT. Caixa Postal 6487 LUANDA Tel: 35515,35516 E P

#### ARGENTINA

Hewlett-Packard Argentina S.A. Avenida Santa Fe 2035 Martinez 1640 BUENOS AIRES Tel: 798-5735, 792-1293 Telex: 17595 BIONAR Cable: HEWPACKARG A,E,CH,CS,P Biotron S.A.C.I.M. e I. Av Paseo Colon 221, Piso 9 1399 BUENOS AIRES Tel: 30-4846, 30-1851 Telex: 17595 BIONAR M

## AUSTRALIA

Adeiaide, South Australia Office Hewlett-Packard Australia Ltd. 153 Greenhil Road PARKSIDE, S.A. 5063 Tel: 272-5911 Telex: 82536 Cable: HEWPARD Adeiaide A\*,CH,CM,,E,MS,P

#### Brisbane, Queensland Office

Hewlett-Packard Australia Ltd. 10 Payne Road THE GAP, Queensland 4061 Tel: 30-4133 Telex: 42133 Cable: HEWPARD Brisbane A,CH,CM,E,M,P Canberra, Australia

## Capitai Territory

Office Hewlett-Packard Australia Ltd. 121 Wolkongong Street FYSHWICK, A.C.T. 2609 Tel: 80 4244 Telex: 62650 Cable: HEWPARD Canberra CH,CM,E,P

#### Melbourne, Victoria Office Hewlett-Packard Australia Ltd.

31-41 Joseph Street BLACKBURN, Victoria 3130 Tel: 895-2895 Telex: 31-024 Cable: HEWPARD Melbourne ACH,CM,CS,E,MS,P

## Perth, Western Australia Office Hewleti-Packard Australia Ltd.

261 Stirling Highway CLAREMONT, W.A. 6010 Tel: 383-2188 Telex: 93859 Cable: HEWPARD Perth A.CH.CM.E.MS.P

#### Sydney, New South Wales Office

Hewlett-Packard Australia Ltd. 17-23 Talavera Road P.O. Box 308 NORTH RYDE, N.S.W. 2113 Tel: 887-1611 Telex: 21561 Cable: HEWPARD Sydney A.CH.CM.CS.E.MS.P

#### AUSTRIA

Hewleti-Packard Ges.m.b.h. Grottenhofstrasse 94 A-8052 GRAZ Tel: (0316) 291 5 66 Telex: 32375 CH,E Hewleti-Packard Ges.m.b.h. Lieblgasse 1 P.O. Box 72 A-1222 VIENNA Tel: (0222) 23 65 11-0 Telex: 13425 HEPA A A,CH,CM,CS,E,MS,P

#### BAHRAIN

Green Salon P.O. Box 557 Manama BAHRAIN Tel: 255503-255950 Telex: 84419

Wael Pharmacy P.O. Box 648 BAHRAIN Tel: 256 123 Telex: 8550 WAEL BN E.C.M

## BELGIUM

Hewlett-Packard Belgium S.A./N.V. Bivd de la Woluwe, 100 Woluwedal B-1200 BRUSSELS Tel: (02) 762-32-00 Telex: 23-494 paloben bru A.CH.CM.CS.E.MP.P

#### BRAZIL

Hewlett-Packard do Brasil I.e.C. Ltda. Alameda Rio Negro, 750 Alphaville 06400 BARUERI SP Tel: (011) 421.1311 Telex: (011) 33872 HPBR-BR Cable: HEWPACK Sao Paulo A,CH,CM,CS,E,M,P Hewlett-Packard do Brasil I.e.C. Ltda. Avenida Epitacio Pessoa, 4664 22471 RIO DE JAMEIRO-RJ Tel: (021) 266.0237

Telex: 021/250.0257 Telex: 021-21905 HPBR-BR Cable: HEWPACK Rio de Janeiro A,CH,CM,E,MS,P\* ANAMED I.C.E.I. Ltda Rua Bage, 103 04012 SAO PAULO Tel: (011) 570-5726 Telex: 021-21905 HPBR-BR





# **SALES & SUPPORT OFFICES**

Arranged alphabetically by country

## CANADA

Alberta Hewlett-Packard (Canada) Ltd. 3030 3rd Avenue N.E. CALGARY, Alberta T2A 6T7 Tel: (403) 235-3100 A,CH,CM,E\*,MS,P\* Hewlett-Packard (Canada) Ltd. 11120A-178th Street EDMONTON, Alberta T5S 1P2 Tel: (403) 486-6666

#### A,CH,CM,CS,E,MS,P British Columbia Hewlett-Packard (Canada) Ltd. 10691 Shelibridge Way RICHMOND, British Columbia V6X 2W7 Tel: (604) 270-2277

Tel: (604) 270-2277 Telex: 610-922-5059 A,CH,CM,CS,E\*,MS,P\*

Manitoba Hewlett-Packard (Canada) Ltd. 380-550 Century Street WINNIPEG, Manitoba R3H 0Y1 Tel: (204) 786-6701 A,CH,CM,E,MS,P\*

#### Nova Scotia Hewlett-Packard (Canada) Ltd. P.O. Box 931 900 Windmill Road DARTMOUTH, Nova Scotia B2Y 326 Tel: (902) 469-7820 CH,CM,CS,E\*,MS,P\*

#### Ontario

Hewlett-Packard (Canada) Ltd. 3325 N. Service Rcl., Unit 6 BURLINGTON, Ontario P3A 2A3 Tel: (416) 335-8644 CS,M\*

Hewlett-Packard (Canada) Ltd. 552 Newbold Street LONDON, Ontario N6E 2S5 Tel: (519) 686-9181 A,CH,CM,E\*,MS,P\*

Hewlett-Packard (Canada) Ltd. 6877 Goreway Drive MISSISSAUGA, Ontario L4V 1M8 Tel: (416) 678-9430 A,CH,CM,CS,E,MP,P Hewlett-Packard (Canada) Ltd. 2670 Queensview Dr. OTTAWA, Ontario K2B 8K1 Tel: (613) 820-6483

A,CH,CM,CS,E\*,MS,P\* Hewlett-Packard (Canada) Ltd. 220 Yorkland Blvd., Unit #11 WILLOWDALE, Ontario M2J 1R5 Tel: (416) 499-9333 CH

## Quebec

Hewlett-Packard (Canada) Ltd. 17500 South Service Road Trans-Canada Highway KIRKLAND, Ouebec H9J 2M5 Tel: (514) 697-4232 A,CH,CM,CS,E,MP,P\*

Hewlett-Packard (Canada) Ltd. Les Galeries du Vallon 2323 Du Versont Nord STE. FOY, Quebec G1N 4C2 Tel: (418) 687-4570 CH

## CHILE

Jorge Calcagni y Cia. Ltda. Av. Italia 634 Santiago Casilla 16475 SANTIAGO 9 Tel: 222-0222 Telex: Public Booth 440001 A.CM.E.M Olympia (Chile) Ltda. Av. Rodrigo de Araya 1045 Casilla 256-V SANTIAGO 21 Tel: (02) 22 55 044 Telex: 240-565 04 YMP CL Cable: Olympiachile Santiagochile CH,CS,P

#### CHINA, People's Republic of China Hewlett-Packard Rep. Office P.O. Box 418 1A Lane 2, Luchang St. Beiwei Rd., Xuanwu District BELUNG

Tel: 33-1947, 33-7426 Telex: 22601 CTSHP CN Cable: 1920 A,CH,CM,CS,E,P

#### COLOMBIA Instrumentación

H. A. Langebaek & Kier S.A. Carrera 4A No. 52A-26 Apartado Aereo 6287 BOGOTA 1, D.E. Tel: 212-1466 Telex: 44400 INST CO Cable: AARIS Bogota CM.E.M Casa Humboldt Lida. Carrera 14, No. 98-60 Apartado Aereo 51283 BOGOTA 1, D.E. Tel: 256-1686 Telex: 45403 CCAL CO. A

## **COSTA RICA**

Cientifica Costarricense S.A. Avenida 2, Calle 5 San Pedro de Montes de Oca Apartado 10159 SAN JOSÉ Tel: 24-38-20, 24-08-19 Telex: 2367 GALGUR CR CM.EM

#### CYPRUS Telerexa Ltd. P.O. Box 4809 14C Stassinos Avenue NICOSIA Tel: 62698 Telex: 2894 LEVIDO CY

EMP

DENMARK Hewlett-Packard A/S Datavej 52 DK-3460 BIRKEROD Tel: (02) 81-66-40 Telex: 37409 hpas dk A,CH,CM,CS,E,MS,P Hewlett-Packard A/S Rolighedsvej 32 DK-8240 RISSKOV, Aarhus Tel: (06) 17-60-00 Telex: 37409 hpas dk CH,E

## DOMINICAN REPUBLIC

Microprog S.A. Juan Tomás Mejía y Coles No. 60 Arroyo Hondo **SANTO DOMINGO** Tel: 565-6268 Telex: 4510 ARENTA DR (RCA) P

ECUADOR CYEDE Cia. Ltda. Avenida Eloy Altaro 1749 Casilia 6423 CCI QUITO Tel: 450-975, 243-052 Telex: 2548 CYEDE ED CM,E,P Hospitalar S.A. Robles 625 Casilia 3590 QUITO Tel: 545-250, 545-122 Telex: 2485 HOSPTL ED Cable: HOSPITALAR-Quito

# EGYPT

International Engineering Associates 24 Hussein Hegazi Street Kasr-el-Ahi CAMO Tel: 23829, 21641 Telex: IEA UN 93830 CH,CS,E,M EGYPOR P.O.Box 2558 42 El Zahraa Street CAMO, Egypt Tel: 65 00 21 Telex: 93 337

#### EL SALVADOR IPESA de El Salvador S.A. 29 Avenida Norie 1216 SAN SALVADOR Tal: 26-6858, 26-6868 Telex: 20539 IPESASAL A.CLCM.CS.E.P

FINLAND Hewlett-Packard Oy **Revontulentie 7** PL 24 SF-02101 ESPO0 10 Tel: (90) 4550211 Telex: 121563 hewpa sf CH,CM,CS,P Hewlett-Packard Oy (Olarinluoma 7) PL 24 02101 ESPO0 10 Tel: (90) 4521022 A,E,MS Hewlett-Packard Oy Aatoksenkatv 10-C SF-40720-72 JYVASKYLA Tel: (941) 216318 CH

Hewlett-Packard Oy Kainvuntie 1-C SF-90140-14 OULU Tel: (981) 338785 CH

#### FRANCE

Hewlett-Packard France Z.I. Mercure B Rue Berthelot F-13763 Les Milles Cedex AIX-EN-PROVENCE Tel: 16 (42) 59-41-02 Telex: 410770F A,CH,E,MS,P Hewlett-Packard France 64, rue Marchand Saillant F-61000 ALENCON Tel: 16 (33) 29 04 42 Hewlett-Packard France **Boite Postale 503** F-25026 BESANCON 28 rue de la Republique F-25000 BESANCON Tel: 16 (81) 83-16-22 CH,M Hewlett-Packard France 13, Place Napoleon III F-29000 BREST Tel: 16 (98) 03-38-35 Hewlett-Packard France **Chemin des Mouilles** Boite Postale 162 F-69130 ECULLY Cedex (Lyon) Tel: 16 (78) 833-81-25 Telex: 310617F A.CH.CS.E.MP Hewlett-Packard France **Tour Lorraine Boulevard de France** F-91035 EVRY Cedex Tel: 16 6 077-96-60 Telex: 692315F F Hewlett-Packard France Parc d'Activite du Bois Briard Ave. du Lac F-91040 EVRY Cedex Tel: 16 6 077-8383 Telex: 692315F Hewlett-Packard France 5, avenue Raymond Chanas F-38320 EYBENS (Grenoble) Tel: 16 (76) 25-81-41 Telex: 980124 HP GRENOB EYBE СН Hewlett-Packard France Centre d'Affaire Paris-Nord Bâtiment Ampère 5 étage Rue de la Commune de Paris **Boite Postale 300** F-93153 LE BLANC MESNIL Tel: 16 (1) 865-44-52 Telex: 211032F CH.CS.E.MS Hewlett-Packard France Parc d'Activités Cadera Quartier Jean Mermoz Avenue du Président JF Kennedy F-33700 MERIGNAC (Bordeaux) Tel: 16 (56) 34-00-84 Telex: 550105F CH.E.MS Hewlett-Packard France immueble "Les 3 B" Nouveau Chemin de la Garde ZAC de Bois Briand F-44085 NANTES Cedex Tel: 16 (40) 50-32-22 CH..

FRANCE (Cont'd) Hewlett-Packard France 125, rue du Faubourg Bannier F-45000 ORLEANS Tel: 16 (38) 68 01 63 Hewlett-Packard France Zone Industrielle de Courtaboeuf Avenue des Tropiques F-91947 Les Ulis Cedex ORSAY Tel: (6) 907-78-25 Telex: 600048F A,CH,CM,CS,E,MP,P Hewlett-Packard France Paris Porte-Maillot 15, Avenue de L'Amiral Bruix F-75782 PARIS CEDEX 16 Tel: 16 (1) 502-12-20 Telex: 613663F CH.MS.P **Hewlett-Packard France** 124, Boulevard Tourasse F-64000 PAU Tel: 16 (59) 80 38 02 Hewlett-Packard France 2 Allée de la Bourgonnette F-35100 RENNES Tel: 16 (99) 51-42-44 Telex: 740912F CH.CM.E.MS.P\* Hewlett-Packard France 98 Avenue de Bretagne F-76100 ROUEN Tel: 16 (35) 63-57-66 CH\*\*,CS Hewlett-Packard France 4 Rue Thomas Mann **Boite Postale 56** F-67033 STRASBOURG Cedex Tel: 16 (88) 28-56-46 Telex: 890141F CH.E.MS.P\* **Hewlett-Packard France** Le Péripole 20, Chemin du Pigeonnier de la Cépière F-31083 TOULOUSE Cedex Tel: 16 (61) 40-11-12 Telex: 531639F A,CH,CS,E,P\* **Hewlett-Packard France** 9, rue Baudin F-26000 VALENCE Tel: 16 (75) 42 76 16 Hewlett-Packard France Carolor ZAC de Bois Briand F-57640 VIGY (Metz) Tel: 16 (8) 771 20 22 СН Hewlett-Packard France Immeuble Péricentre F-59658 VILLENEUVE D'ASCO Cedex Tel: 16 (20) 91-41-25 Telex: 160124F CH,E,MS,P\*

#### GERMAN FEDERAL REPUBLIC

Hewlett-Packard GmbH Geschäftsstelle Keithstrasse 2-4 D-1000 BERLIN 30 Tel: (030) 24-90-86 Telex: 018 3405 hpbln d A.CH.E.M.P

Hewlett-Packard GmbH Geschäftsstelle Herrenberger Strasse 130 D-7030 BÖBLINGEN Tel: (7031) 14-0 Telev A,CH,CM,CS,E,MP,P Hewlett-Packard GmbH Geschäftsstelle Emanuel-Leutze-Strasse 1 D-4000 DUSSELDORF Tel: (0211) 5971-1 Telex: 085/86 533 hpdd d A.CH.CS.E.MS.P Hewlett-Packard GmbH Geschäftsstelle Schleefstr. 28a D-4600 DORTMUND-Apierbeck Tei: (0231) 45001 Hewlett-Packard GmbH Vertriebszentrale Frankfurt Berner Strasse 117 Postfach 560 140 D-6000 FRANKFURT 56 Tel: (0611) 50-04-1 Telex: 04 13249 hpffm d A,CH,CM,CS,E,MP,P Hewlett-Packard GmbH Geschäftsstelle Aussenstelle Bad Homburg Louisenstrasse 115 D-6380 BAD HOMBURG Tel: (06172) 109-0 Hewlett-Packard GmbH Geschäftsstelle Kapstadtring 5 D-2000 HAMBURG 60 Tel: (040) 63804-1 Telex: 021 63 032 hphh d A,CH,CS,E,MS,P Hewlett-Packard GmbH Geschäftsstelle Heidening 37-39 D-3000 HANNOVER 61 Tel: (0511) 5706-0 Telex: 092 3259 A,CH,CM,E,MS,P Hewlett-Packard GmbH Geschäftsstelle Rosslauer Weg 2-4 D-6800 MANNHEIM Tel: (0621) 70050 Telex: 0462105 A,C,E Hewlett-Packard GmbH Geschäftsstelle Messerschmittstrasse 7 D-7910 NEU ULM Tel: 0731-70241 Telex: 0712816 HP ULM-D A,C.E Hewlett-Packard GmbH Geschäftsstelle Ehhericherstr. 13 D-8500 NÜRNBERG 10 Tel: (0911) 5205-0 Telex: 0623 860 CH,CM,E,MS,P Hewieti-Packard GmbH Geschäftsstelle Eschenstrasse 5 D-8028 TAUFKIRCHEN Tel: (089) 6117-1 Telex: 0524985

A.CH.CM.E.MS.P

# SALES & SUPPORT OFFICES

Arranged alphabetically by country

# GREAT BRITAIN

## See United Kingdom

GREECE Kostas Karaynnis S.A. 8 Omirou Street ATHENS 133 Tel: 32 30 303, 32 37 371 Telex: 215962 RKAR GR A,CH,CM,CS,E,M,P PLAISIO S.A. G. Gerardos 24 Stourmara Street ATHENS Tel: 36-11-160 Telex: 221871

## GUATEMALA

IPESA Avenida Reforma 3-48, Zona 9 GUATEMALA CITY Tel: 316627, 314786 Telex: 4192 TEL TRO GU ACHCM (SEMP)

## HONG KONG

Hewlett-Packard Hong Kong, Ltd. G.P.O. Box 795 5th Floor, Sun Hung Kai Centre 30 Harbour Road HONG KONG Tel: 5-8323211 Telex: 66678 HEWPA HX Cable: HEWPACK HONG KONG E.CH.CS.P CET LId. 1402 Tung Wah Mansion 199-203 Hennessy Rd. Wanchia, HONG KONG Tel: 5-729376 Telex: 85148 CET HX CM Schmidt & Co. (Hong Kong) Ltd. Wing On Centre, 28th Floor Connaught Road, C. HONG KONG Tel: 5-455644 Telex: 74766 SCHMX HX A M

## ICELAND

Elding Trading Company Inc. Hafnamvoli-Tryggvagotu P.O. Box 895 IS-REYKJAVIK Tel: 1-58-20, 1-63-03 M

#### INDIA

Computer products are sold through Blue Star Ltd. All computer repairs and maintenance service is done through Computer Maintenance Corp. Blue Star Ltd. Sabri Complex II Floor 24 Residency Rd. BANGALORE 560 025 Tel: 55660 Teles: 0845-430 Cable: BLUESTAR A,CH\*,CM,CS\*,E

#### Blue Star Ltd. Band Box House Prahhadevi BOMBAY 400 025 Tel: 422-3101 Telex: 011-3751 Cable: BLUESTAR A 14 Blue Star Ltd. Sahas 414/2 Vir Savarkar Marg Prabhadevi BOMBAY 400 025 Tel: 422-6155 Telex: 011-4093 Cable: FROSTBLUE A,CH\*,CM,CS\*,E,M Blue Star Ltd. Kalyan, 19 Vishwas Colony Alkapuri, BORODA, 390 005 Tel: 65235 Cable: BLUE STAR Blue Star Ltd. 7 Hare Street CALCUTTA 700 001 Tet: 12-01-31 Telex: 021-7655 Cable: BLUESTAR A.M Blue Star Ltd. 133 Kodambakkam High Road MADRAS 600 034 Tel: 82057 Telex: 041-379 Cable: BLUESTAR A M Blue Star Ltd. Bhandari House, 7th/8th Floors 91 Nehru Place NEW DELHI 110 024 Tel: 682547 Telex: 031-2463 Cable: BLUESTAR A,CH\*,CM,CS\*,E,M Blue Star Ltd. 15/16:C Wellesley Rd. PUNE 411 011

Tel: 22775 Cable: BLUE STAR Blue Star Ltd. 2-2-47/1108 Bolarum Rd. SECUNDERABAD 500 003 Tel: 72057 Telex: 0155-459 Cable: BLUEFROST A.E Blue Star Ltd. T.C. 7/603 Poornima Maruthankuzhi TRIVANDRUM 695 013 Tel: 65799 Telex: 0884-259 Cable: BLUESTAR Computer Maintenance Corporation Ltd 115, Sarojini Devi Road SECUNDERABAD 500 003 Tel: 310-184, 345-774 Telex: 031-2960

CH··





# SALES & SUPPORT OFFICES

## Arranged alphabetically by country

**INDONESIA** BERCA Indonesia P.T. P.O.Box 496/Jkl. JI. Abdul Muis 62 JAKARTA Tel: 21-373009 Telex: 46748 BERSAL IA Cable: BERSAL JAKARTA BERCA Indonesia P.T. P.O.Box 2497/Jkt Antara Bidg., 17th Floor Jl. Medan Merdeka Selatan 17 JAKARTA-PUSAT Tel: 21-344-181 Telex: BERSAL IA A,CS,E,M BERCA Indonesia P.T. P.O. Box 174/SBY. JI. Kulei No. 11 SURABAYA Tel: 68172 Telex: 31146 BERSAL SB Cable: BERSAL-SURABAYA A'EMP

## IRAQ

Hewlett-Packard Trading S.A. Service Operation Al Mansoor City 9B/3/7 BAGHDAD Tel: 551-49-73 Telex: 212-455 HEPAIRAQ IK CH,CS

## IRELAND Hewlett-Packard Ireland Ltd.

82/83 Lower Leeson Street DUBLIN 2 Tel: 0001 608800 Telex: 30439 A,CH,CM,CS,E,M,P Cardiac Services LId. Kilmore Road Arlane DUBLIN 5 Tel: (01) 351820 Telex: 30439

### ISRAEL

Eldan Electronic Instrument Ltd. P.O.Box 1270 JERUSALEM 91000 16, Ohaliav St. JERUSALEM 94467 Tel: 533 221, 553 242 Telex: 25231 AB/PAKRD IL Electronics Engineering Division Motorola Israel Ltd. 16 Kremenetski Street P.O. Box 25016 TEL-AVIV 67899 Tel: 3 88 388 Telex: 33569 Motil IL Cable: BASTEL Tel-Aviv CH,CM,CS,E,M,P

## ITALY

Hewlett-Packard Italiana S.p.A Traversa 99C Via Giulio Petroni, 19 I-70124 **BARI** Tei: (080) 41-07-44 M Hewlett-Packard Italiana S.p.A. Via Martin Luther King, 38/III I-40132 BOLOGNA Tel: (051) 402394 Telex: 511630 CH,E,MS Hewlett-Packard Italiana S.p.A. Via Principe Nicola 43G/C 1-95126 CATANIA Tel: (095) 37-10-87 Telex: 970291 CP Hewlett-Packard Italiana S.p.A. Via G. Di Vittorio 9 1-20063 CERNUSCO SUL NAVIGLIO (Milano) Tel: (02) 923691 Telex: 334632 A,CH,CM,CS,E,MP,P Hewlett-Packard Italiana S.p.A. Via C. Colombo 49 I-20090 TREZZANO SUL NAVIGLIO (Milano) Tel: (02) 4459041 Telex: 322116 C,M Hewlett-Packard Italiana S.p.A. Via Nuova San Rocco a Capodimonte, 62/A 1-80131 NAPOLI Tel: (081) 7413544 Telex: 710698 A,CH,E Hewlett-Packard Italiana S.p.A. Viale G. Modugno 33 I-16156 GENOVA PEGLI Tel: (010) 68-37-07 Telex: 215238 E.C Hewlett-Packard Italiana S.p.A. Via Pelizzo 15 1-35128 PADOVA Tel: (049) 664888 Telex: 430315 A,CH,E,MS Hewletl-Packard Italiana S.p.A. Viale C. Pavese 340 1-00144 ROMA EUR Tel: (06) 54831 Telex: 610514 A,CH,CM,CS,E,MS,P\* Hewlett-Packard Italiana S.p.A. Via di Casellina 57/C I-50018 SCANDICCI-FIRENZE Tel: (055) 753863 Hewlett-Packard Italiana S.p.A. Corso Svizzera, 185 I-10144 TORINO Tel: (011) 74 4044 Telex: 221079 CH,E JAPAN

Yokogawa-Hewlett-Packard Ltd. 152-1, Onna ATSUGI, Kanagawa, 243 Tel: (0462) 28-0451 CM,C\*,E Yokogawa-Helwett-Packard Ltd. Meiji-Seimei Bldg. 6F 3-1 Hon Chiba-Cho CHIBA, 280 Tel: 472 25 7701 E,CH,CS

Yokogawa-Hewlett-Packard Ltd. Yasuda-Seimei Hiroshima Bldg. 6-11, Hon-dori, Naka-ku HIROSHIMA, 730 Tel: 82-241-0611 Yokogawa-Hewleit-Packard Ltd. Towa Building 2-3, Kaigan-dori, 2 Chome Chuo-ku KOBE, 650 Tel: (078) 392-4791 C.E Yokogawa-Hewlett-Packard Ltd. Kumagaya Asahi 82 Bldg 3-4 Tsukuba KUMAGAYA, Saitama 360 Tel: (0485) 24-6563 CH,CM,E Yokogawa-Hewlett-Packard Ltd. Asahi Shinbun Daiichi Seimei Bldg. 4-7, Hanabata-cho KUMAMOTO,860 Tel: (0963) 54-7311 CH.E Yokogawa-Hewlett-Packard Ltd. Shin-Kyoto Center Bldg. 614, Higashi-Shiokoji-cho Karasuma-Nishiiru Shiokoji-dori, Shirnogyo-ku KYOTO, 600 Tel: 075-343-0921 CH,E Yokogawa-Hewlett-Packard Ltd. Mito Mitsui Bldg 4-73, Sanno-maru, 1 Chome MITO, Ibaraki 310 Tel: (0292) 25-7470 CH,CM,E Yokogawa-Hewlett-Packard Ltd. Sumitomo Seimei 14-9 Bldg. Meieki-Minami, 2 Chome Nakamura-ku NAGOYA, 450 Tel: (052) 571-5171 CH,CM,CS,E,MS Yokogawa-Hewlett-Packard Ltd. Chuo Bldg., 4-20 Nishinakajima, 5 Chome Yodogawa-ku **OSAKĂ**, 532 Tel: (06) 304-6021 Telex: YHPOSA 523-3624 A,CH,CM,CS,E,MP,P Yokogawa-Hewlett-Packard Ltd. 27-15, Yabe, 1 Chome SAGAMIHARA Kanagawa, 229 Tel: 0427 59-1311 Yokogawa-Hewlett-Packard Ltd. Daiichi Seimei Bldg. 7-1, Nishi Shinjuku, 2 Chome Shinjuku-ku, TOKYO 160 Tel: 03-348-4611 CH.E Yokogawa-Hewlett-Packard Ltd. 29-21 Takaido-Higashi, 3 Chome Suginami-ku TOKYO 168 Tel: (03) 331-611 Telex: 232-2024 YHPTOK A.CH.CM.CS.E.MP.P\* Yokogawa-Hewlett-Packard Ltd. Daiichi Asano Building 2-8, Odori, 5 Chome UTSUNOMIYA, Tochigi 320 Tel: (0286) 25-7155 CH,CS,E

Yokogawa-Hewlett-Packard Ltd. Yasuda Seimei Nishiguchi Bldg. 30-4 Tsuruya-cho, 3 Chome YOKOHAMA 221 Tel: (045) 312-1252 CH,CM,E

#### JORDAN

Mouasher Cousins Company P.O. Box 1387 AMMAN Tel: 24907, 39907 Telex: 21456 SABCO JO CH F M P

#### **KENYA**

ADCOM Ltd., Inc., Kenya P.O.Box 30070 NAIROBI Tel: 331955 Telex: 22639 E.M

### KOREA

Samsung Electronics HP Division 12 Fl. Kinam Bldg. San 75-31, Yeoksam-Dong Kangnam-Ku Yeongdong P.O. Box 72 SEOUL Tel: 555-7555, 555-5447 Telex: K27364 SAMSAN A,CH,CM,CS,E,M,P

#### KUWAIT

Al-Khaldiya Trading & Contracting P.O. Box 830 Safat KUWAIT Tel: 42-4910, 41-1726 Telex: 22481 Areeg kt CH.E.M Photo & Cine Equipment P.O. Box 270 Safat KUWAIT Tel: 42-2846, 42-3801 Telex: 22247 Matin kt P

#### LEBANON

G.M. Dolmadjian Achrafieh P.O. Box 165.167 BEIRUT Tel: 290293 MP\*\* Computer Information Systems P.O. Box 11-6274 BEIRUT Tel: 89 40 73 Telex: 22259

#### LUXEMBOURG

Hewlett-Packard Belgium S.A./N.V. Blvd de la Woluwe, 100 Woluwedal B-1200 **BRUSSELS** Tel: (02) 762-32-00 Telex: 23-494 paloben bru A,CH,CM,CS,E,MP,P

#### MALAYSIA

Hewlett-Packard Sales (Malaysia) Sdn. Bhd. 1st Floor, Bangunan British American Jalan Semantan, Damansara Heights KUALA LUMPUR 23-03 Tel: 943022 Telex: MA31011 A,CH,E,M,P\* MAYLAYSIA (Cont'd) Protel Engineering P.O.Box 1917 Lot 6624, Section 64 23/4 Pending Road Kuching, SARAWAK Tel: 36299 Telex: MA 70904 PROMAL Cable: PROTELENG A.E.M

MALTA Philip Toledo Lid. Notabile Rd. MRIEMEL Tel: 447 47, 455 66 Telex: Media MW 649 E,P

MEXICO Hewlett-Packard Mexicana, S.A. de C.V. Av. Periferico Sur No. 6501 Tepepan, Xochimilco 16020 MEXICO D.F. Tel: 6-76-46-00 Telex: 17-74-507 HEWPACK MEX A,CH,CS,E.MS.P Hewlett-Packard Mexicana, S.A. de C.V. Ave. Colonia del Valle 409 Col. del Valle Municipio de Garza Garcia MONTERREY, NUEVO LEON Tel: 78 42 41 Telex: 038 410 СН ECISA José Vasconcelos No. 218 Col. Condesa Deleg. Cuauhtémoc MEXICO D.F. 06140 Tel: 553-1206 Telex: 17-72755 ECE ME

#### MOROCCO

Dolbeau 81 rue Karatchi CASABLANCA Tel: 3041-82, 3068-38 Telex: 23051, 22822 5

Gerep 2 rue d'Agadir Boite Postale 156 CASABLANCA Tel: 272093, 272095 Telex: 23 739

#### NETHERLANDS

Hewlett-Packard Nederland B.V. Van Heuven Goedharttaan 121 NL 1181KK AMSTELVEEN P.O. Box 667 NL 1180 AR AMSTELVEEN Tel: (020) 47-20-21 Telex: 13 216 HEPA NL A,CH,CM,CS,E,MP,P Hewlett-Packard Nederland B.V. Bongerd 2 NL 2906VK CAPELLE A/D IJSSEL

NL 2900VK CAPELLE A/D INSSEL P.O. Box 41 NL 2900AA CAPELLE A/D INSSEL Tel: (10) 51-64-44 Tel:x: 21261 HEPAC NL A.CH.CS.E Hewlett-Packard Nederland B.V. Pastoor Petersstraat 134-136 NL 5612 LV EINDHOVEN P.O. Box 2342 NL 5600 CH EINDHOVEN Tel: (040) 326911 Telex: 51484 hepae nl A.CH\*\*E.M

#### NEW ZEALAND

Hewlett-Packard (N.Z.) Ltd. 5 Owens Road P.O. Box 26-189 Epsom, AUCKLAND Tel: 687-159 Cable: HEWPACK Auckland CH,CM,E,P\* Hewlett-Packard (N.Z.) Ltd. 4-12 Cruickshank Street Kilbirnie, WELLINGTON 3 P.O. Box 9443 Courtenay Place, WELLINGTON 3 Tel: 877-199 Cable: HEWPACK Wellington CH,CM,E,P Northrop Instruments & Systems Ltd. 369 Khyber Pass Road P.O. Box 8602 AUCKLAND Tel: 794-091 Telex: 60605 A,M Northrop Instruments & Systems Ltd. 110 Mandeville SI. P.O. Box 8388 CHRISTCHURCH Tel: 486-928 Telex: 4203 A,M Northrop Instruments & Systems Ltd. Sturdee House 85-87 Ghuznee Street P.O. Box 2406 WELLINGTON Tel: 850-091 Telex: NZ 3380 A.M

## NORTHERN IRELAND See United Kingdom

NORWAY

Hewlett-Packard Norge A/S Folke Bernadottes vei 50 P.O. Box 3558 N-5033 FYLLINGSDALEN (Bergen) Tel: 0047/5/16 55 40 Telex: 16621 hpnas n CH,CS,E,MS Hewlett-Packard Norge A/S Österndalen 16-18 P.O. Box 34 N-1345 ÖSTERÅS Tel: 0047/2/17 11 80 Telex: 16621 hpnas n A,CH,CM,CS,E,M,P

## OMAN

Khinjil Ramdas P.O. Box 19 MUSCAT Tel: 722225, 745601 Telex: 3289 BROKER MB MUSCAT P Suhail & Saud Bahwan P.O.Box 169 MUSCAT Tel: 734 201-3 Telex: 3274 BAHWAN MB SALES & SUPPORT OFFICES

Arranged alphabetically by country

#### PAKISTAN

Mushko & Company Lid. 1-B, Street 43 Sector F-8/1 ISLAMABAD Tel: 51071 Cable: FEMUS Rawalpindi A,E,M Mushko & Company Lid. Oosman Chambers Abdullah Haroon Road KARACH 0302 Tel: 524131, 524132 Telex: 2894 MUSKO PK Cable: COOPERATOR Karachi A,E,M,P\*

## PANAMA

Electrónico Balboa, S.A. Calle Samuel Lewis, Ed. Alfa Apartado 4929 **PANAMA 5** Tel: 63-6613, 63-6748 Telex: 3483 ELECTRON PG A,CM,E,M,P

#### PERU

Cía Electro Médica S.A. Los Flamencos 145, San Isidro Casilia 1030 LIMA 1 Tel: 41-4325, 41-3703 Telex: Pub. Booth 25306 CM.E.M.P

## PHILIPPINES

The Online Advanced Systems Corporation Rico House, Amorsolo Cor. Herrera Street Legaspi Village, Makati P.O. Box 1510 Metro MANILA Tel: 85-35-81, 85-34-91, 85-32-21 Telex: 3274 ONLINE A.CH.CS.E.M Electronic Specialists and Proponents Inc. 690-B Epifanio de los Santos Avenue Cubao, QUEZON CITY P.O. Box 2649 Manila Tel: 98-96-81, 98-96-82, 98-96-83 Telex: 40018, 42000 ITT GLOBE MACKAY BOOTH

## PORTUGAL

Mundinter Intercambio Mundial de Comércio S.A.R.L. P.O. Box 2761 Av. Antonio Augusto de Aguiar 138 P-LI**SBON** Tel: (19) 53-21-31, 53-21-37 Telex: 16691 munter p M Soquimica

Soquinica Av. da Liberdade, 220-2 1298 LI**SBOA** Codex Tel: 56 21 81/2/3 Telex: 13316 SABASA

Telectra-Empresa Técnica de Equipmentos Eléctricos S.A.R.L. Rua Rodrigo da Fonseca 103 P.O. Box 2531 P-LISBON 1 Tel: (19) 68-60-72 Telex: 12598 CH.CS,E,P

## PUERTO RICO

Hewlett-Packard Puerto Rico Ave. Muñoz Rivera #101 Esq. Calle Ochoa NATO REY, Puerto Rico 00918 Tel: (809) 754-7800 Hewlett-Packard Puerto Rico Calle 272 Edificio 203 Urb. Country Club RIO PIEDRAS, Puerto Rico P.O. Box 4407 CAROLINA, Puerto Rico 00628 Tel: (809) 762-7255 A,CH,CS

## QATAR

Computearbia P.O. Box 2750 DOHA Tel: 883555 Telex: 4806 CHPARB P Eastern Technical Services P.O.Box 4747 DOHA Tel: 329 993 Telex: 4156 EASTEC DH Nasser Trading & Contracting P.O.Box 1563 DOHA Tel: 22170, 23539 Telex: 4439 NASSER DH

## SAUDI ARABIA

Modern Electronic Establishment Hewlett-Packard Division P.O. Box 22015 Thuobah AL-KHOBAR Tel: 895-1760, 895-1764 Telex: 671 106 HPMEEK SJ Cable: ELECTA AL-KHOBAR CH.CS.E.M Modern Electronic Establishment Hewlett-Packard Division P.O. Box 1228 Redec Plaza, 6th Floor JEDDAH Tel: 644 38 48 Telex: 4027 12 FARNAS SJ Cable: ELECTA JEDDAH CH.CS.E.M Modern Electronic Establishment Hewlett-Packard Division P.O.Box 22015 RIYADH Tel: 491-97 15, 491-63 87 Telex: 202049 MEERYD SJ CH.CS.E.M Abdul Ghani El Ajou P.O. Box 78 RIYADH Tel: 40 41 717 Telex; 200 932 EL AJOU

#### SCOTLAND

## See United Kingdom

SINGAPORE Hewlett-Packard Singapore (Sales) Pte. Ltd. #08-00 Inchcape House 450-2 Alexandra Road P.O. Box 58 Alexandra Rd. Post Office SINGAPORE, 9115 Tel: 631788 Telex: HPSGS0 RS 34209 Cable: HEWPACK, Singapore A,CH,CS,E,MS,P





# SALES & SUPPORT OFFICES

## Arranged alphabetically by country

SINGAPORE (Cont'd) Dynamar International Ltd. Unit 05-11 Block 6 Kolam Ayer Industrial Estate SINGAPORE 1334 Tel: 747-6188 Telex: RS 26283 CM

## SOUTH AFRICA

Hewlett-Packard So Africa (Pty.) Ltd. P.O. Box 120 Howard Place CAPE PROVINCE 7450 Pine Park Center, Forest Drive, Pinelands CAPE PROVINCE 7405 Tel: 53-7954 Telex: 57-20006 A,CH,CM,E,MS,P Hewlett-Packard So Africa (Pty.) Ltd. P.O. Box 37099 92 Overport Drive DURBAN 4067 Tel: 28-4178, 28-4179, 28-4110 Telex: 6-22954 CH,CM Hewlett-Packard So Africa (Pty.) Ltd. 6 Linton Arcade 511 Cape Road Linton Grange PORT ELIZABETH 6000 Tel: 041-302148 СН Hewlett-Packard So Africa (Pty.) Ltd. P.O.Box 33345 Glenstantia 0010 TRANSVAAL 1st Floor East Constantia Park Ridge Shopping Centre Constantia Park PRETORIA Tel: 982043 Telex: 32163 CH,E Hewlett-Packard So Africa (Pty.) Ltd. Private Bag Wendywood SANDTON 2144 Tel: 802-5111, 802-5125 Telex: 4-20877 Cable: HEWPACK Johannesburg A,CH,CM,CS,E,MS,P

SPAIN

A,CS,MS,P

Hewlett-Packard Española S.A. Calle Entenza, 321 E-BARCELONA 29 Tel: 322.24.51, 321.73.54 Telex: 52603 hpbee A,CH,CS,E,MS,P Hewlett-Packard Española S.A. Calle San Vicente S/No Edificio Albia II E-BILBAO 1 Tel: 423.83.06 A,CH,E,MS Hewlett-Packard Española S.A. Crta. de la Coruña, Km. 16, 400 Las Rozas E-MADRID Tel: (1) 637.00.11 CH,CS.M Hewlett-Packard Española S.A. Avda. S. Francisco Javier, S/no Planta 10. Edificio Sevilla 2, E-SEVILLA 5 Tel: 64.44.54 Telex: 72933

Hewlett-Packard Española S.A. Calle Ramon Gordillo, 1 (Entlo.3) E-VALENCIA 10 Tel: 361-1354 CH.P

#### SWEDEN

Hewlett-Packard Sverige AB Sunnanvagen 14K S-22226 LUND Tel: (046) 13-69-79 Telex: (854) 17886 (via Spånga office) СН Hewlett-Packard Sverige AB Östra Tuligatan 3 S-21128 MALMÖ Tel: (040) 70270 Telex: (854) 17886 (via Spånga office) Hewlett-Packard Sverige AB Våstra Vintergatan 9 S-70344 ÖREBRO Tel: (19) 10-48-80 Telex: (854) 17886 (via Spånga office) CH Hewlett-Packard Sverige AB Skalholtsgatan 9, Kista Box 19 S-16393 SPÅNGA Tel: (08) 750-2000 Telex: (854) 17886 Telefax: (08) 7527781 A,CH,CM,CS,E,MS,P Hewlett-Packard Sverige AB Frötallisgatan 30 S-42132 VÄSTRA-FRÖLUNDA Tel: (031) 49-09-50 Telex: (854) 17886 (via Spånga office) CH.E.P

## SWITZERLAND

Hewlett-Packard (Schweiz) AG Clarastrasse 12 CH-4058 BASEL Tel: (61) 33-59-20

Hewlett-Packard (Schweiz) AG 7, rue du Bois-du-Lan Case Postale 365 CH-1217 MEYRIN 2 Tel: (0041) 22-83-11-11 Telex:27333 HPAG CH CH,CM,CS Hewlett-Packard (Schweiz) AG Alimend 2 CH-8967 WIDEN Tel: (0041) 57 31 21 11 Telex: 53933 hpag ch Cable: HPAG CH A,CH,CM,CS,E,MS,P

#### SYRIA

General Electronic Inc. Nun Basha Ahnaf Ebn Kays Street P.O. Box 5781 DAMASCUS Tel: 33-24-87 Telex: 411 215 Cable: ELECTROBOR DAMASCUS E Middle East Electronics P.O.Box 2308 Abu Rumnaneh DAMASCUS Tel: 33 4 5 92 Telex: 411 304 M

## TAIWAN

Hewlett-Packard Far East Ltd. Kaohsiung Office 2/F 68-2, Chung Cheng 3rd Road KAOHSIUNG Tel: (07) 241-2318 CH.CS.E Hewlett-Packard Far East Ltd. Taiwan Branch 8th Floor 337 Fu Hsing North Road TAIPEI Tel: (02) 712-0404 Telex: 24439 HEWPACK Cable:HEWPACK Taipei A,CH,CM,CS,E,M,P Ing Lih Trading Co. 3rd Floor, 7 Jen-Ai Road, Sec. 2 TAIPEt 100 Tel: (02) 3948191 Cable: INGLIH TAIPEI

### THAILAND

Unimesa 30 Palpong Ave., Suriwong BANGKOK 5 Tel: 235-5727 Telex: 84439 Simonco TH Cable: UNIMESA Bangkok A,CH,CS,E,M Bangkok Business Equipment Ltd. 5/5-6 Dejo Road BANGKOK Tel: 234-8670, 234-8671 Telex: 87669-BEQUIPT TH Cable: BUS/QUIPT Bangkok

## TRINIDAD & TOBAGO

Caribbean Telecoms Ltd. 50/A Jerningham Avenue P.O. Box 732 **PORT-OF-SPAIN** Tel: 62-44213, 62-44214 Telex: 235,272 HUGCO WG CM,E,M,P

# 

Tunisie Electronique 31 Avenue de la Liberte TUNIS Tel: 280-144 E,P Corema 1 ter. Av. de Carthage TUNIS Tel: 253-821 Telex: 12319 CABAM TN

## TURKEY

Teknim Company Lld. Iran Caddesi No. 7 Kavaklidere, ANKARA Tel: 275800 Telex: 42155 TKNM TR E.M.A. Medina Eldem Sokak No.41/6 Yuksel Caddesi ANKARA Tel: 175 622 Telex: 42 591

## UNITED ARAB EMIRATES

Emitac Ltd. P.O. Box 2711 ABU DHABI Tel: 82 04 19-20 Cable: EMITAC ABUDHABI Emitac Ltd. P.O. Box 1641 SHARJAH Tel: 591 181 Telex: 68136 Emitac Sh CH,CS,E,M,P

## UNITED KINGDOM

GREAT BRITAIN Hewlett-Packard Ltd. Trafalgar House Navigation Road ALTRINCHAM Cheshire WA14 1NU Tel: 061 928 6422 Telex: 668068 A,CH,CS,E,M,MS,P Hewlett-Packard Ltd. Elstree House, Elstree Way BOREHAMWOOD, Herts WD6 1SG Tel: 01 207 5000 Telex: 8952716 E,CH,CS,P Hewlett-Packard Ltd. Oakfield House, Oakfield Grove Clifton BRISTOL, Avon BS8 2BN Tel: 0272 736806 Telex: 444302 CH,CS,E,P Hewlett-Packard Ltd. Bridewell House **Bridewell Place** LONDON EC4V 6BS Tel: 01 583 6565 Telex: 298163 CH,CS,P Hewlett-Packard Ltd. Fourier House 257-263 High Street LONDON COLNEY Herts. AL2 1HA, St. Albans Tel: 0727 24400 Telex: 1-8952716 CH,CS Hewlett-Packard Ltd. Pontefract Road NORMANTON, West Yorkshire WF6 1RN Tel: 0924 895566 Telex: 557355 CH,CS,P Hewlett-Packard Ltd. The Quadrangle 106-118 Station Road **REDHILL, Surrey RH1 1PS** Tel: 0737 68655 Telex: 947234 CH,CS,E,P

Hewlett-Packard Ltd. Avon House 435 Stratford Road Shirley, SOLIHULL, West Midlands 890 4BL Tel: 021 745 8800 Telex: 339105 CH,CS,E,P Hewlett-Packard Ltd. West End House 41 High Street, West End SOUTHAMPTON Hampshire S03 3DQ Tel: 04218 6767 Telex: 477138 CH,CS,P Hewlett-Packard Ltd. Eskdale Rd. Winnersh, WOKINGHAM Berkshire RG11 5DZ Tel: 0734 696622 Telex: 848884 Hewlett-Packard Ltd. King Street Lane Winnersh, WOKINGHAM Berkshire RG11 5AR Tel: 0734 784774 Telex: 847178 A,CH,CS,E,M,MP,P Hewlett-Packard Ltd. Nine Mile Ride Easthampstead, WOKINGHAM Berkshire, 3RG11 3LL Tel: 0344 773100 Telex: 848805 CH.CS.E.P

**GREAT BRITAIN (Cont'd)** 

#### IRELAND

NORTHERN IRELAND Hewlett-Packard Ltd. Cardiac Services Building 95A Finaghy Road South BELFAST BT10 0BY Tel: 0232 625-566 Telex: 747626 CH,CS

SCOTLAND Hewlett-Packard Ltd. SOUTH QUEENSFERRY West Lothian, EH30 9TG Tel: 031 331 1188 Telex: 72682 CH.CM,CS,E,M,P

## UNITED STATES

Alabama Hewlett-Packard Co. 700 Century Park South, Suite 128 BIRMINGHAM, AL 35226 Tel: (205) 822-6802 A,CH,M Hewlett-Packard Co. 420 Wynn Drive HUNTSVILLE, AL 35805 P.O. Box 7700 HUNTSVILLE, AL 35807 Tel: (205) 830-2000 CH,CM,CS,E,M\*

## Arizona

Hewlett-Packard Co. 8080 Pointe Parkway West PHOENIX, AZ 85044 Tel: (602) 273-8000 A,CH,CM,CS,E,MS Hewlett-Packard Co. 2424 East Aragon Road **TUCSON**, AZ 85706 Tel: (602) 889-4631 CH,E,MS\*\* **California** Hewlett-Packard Co.

99 South Hill Dr. BRISBANE, CA 94005 Tel: (415) 330-2500 CHICS Hewlett-Packard Co. P.O. Box 7830 (93747) 5060 E. Clinton Avenue, Suite 102 FRESNO, CA 93727 Tel: (209) 252-9652 CH.CS.MS Hewlett-Packard Co. P.O. Box 4230 1430 East Orangethorpe FULLERTON, CA 92631 Tel: (714) 870-1000 CH,CM,CS,E,MP Hewlett-Packard Co. 320 S. Kellogg, Suite B GOLETA, CA 93117 Tel: (805) 967-3405 CH Hewlett-Packard Co. 5400 W. Rosecrans Boulevard LAWNDALE, CA 90260 P.O. Box 92105 LOS ANGELES, CA 90009 Tel: (213) 970-7500 Telex: 910-325-6608 CH.CM.CS.MP Hewlett-Packard Co. 3155 Porter Oaks Drive PALO ALTO, CA 94304 Tel: (415) 857-8000 CH.CS.E Hewlett-Packard Co. 4244 So. Market Court, Suite A P.O. Box 15976 SACRAMENTO, CA 95852 Tel: (916) 929-7222 A\*.CH.CS.E.MS Hewlett-Packard Co. 9606 Aero Drive P.O. Box 23333 SAN DIEGO, CA 92139 Tel: (619) 279-3200 CH,CM,CS,E,MP Hewlett-Packard Co. 2305 Camino Ramon "C" SAN RAMON, CA 94583 Tel: (415) 838-5900 CH.CS Hewlett-Packard Co. 3005 Scott Boulevard SANTA CLARA, CA 95050 Tel: (408) 988-7000 Telex: 910-338-0586 A CHICMICS E MP Hewiett-Packard Co. 5703 Corsa Avenue WESTLAKE VILLAGE, CA 91362 Tel: (213) 706-6800 E.CH.CS. Colorado

Hewlett-Packard Co. 24 Inverness Place, East ENGLEWOOD, CO 80112 Tel: (303) 649-5000 A,CH,CM,CS,E,MS

# SALES & SUPPORT OFFICES

Arranged alphabetically by country

Connecticut Hewlett-Packard Co. 47 Barnes Industrial Road South P.O. Box 5007 WALLINGFORD, CT 06492 Tel: (203) 265-7801 A.CH.CM.CS.E.MS

Hewlett-Packard Co. 2901 N.W. 62nd Street P.O. Box 24210 FORT LAUDERDALE, FL 33307 Tel: (305) 973-2600 CH,CS,E,MP Hewlett-Packard Co. 6177 Lake Ellenor Drive P.O. Box 13910 ORLANDO, FL 32859 Tel: (305) 859-2900 A,CH,CM,CS,E.MS Hewlett-Packard Co. 5750B N. Hoover Blvd., Suite 123 P.O. Box 15200 TAMPA, FL 33614 Tel: (813) 884-3282 A\*,CH,CM,CS,E\*,M\*

## Georgia

Florida

Hewlett-Packard Co. 2000 South Park Place P.O. Box 105005 ATLANTA, GA 30348 Tel: (404) 955-1500 Telex: 810-766-4890 A,CH,CM,CS,E,MP

#### Hawaii

Hewlett-Packard Co. Kawaiahao Plaza, Suite 190 567 South King Street HONOLULU, HI 96813 Tel: (808) 526-1555 A.CH.E.MS illinois Hewlett-Packard Co. 304 Eldorado Road P.O. Box 1607 BLOOMINGTON, IL 61701 Tel: (309) 662-9411 CH,MS\* Hewlett-Packard Co. 1100 31st Street, Suite 100 DOWNERS GROVE, IL 60515 Tel: (312) 960-5760 CH.CS Hewlett-Packard Co. 5201 Tollview Drive ROLLING MEADOWS, IL 60008 Tel: (312) 255-9800 Telex: 910-687-1066

## A,CH,CM,CS,E,MP Indiana

Hewlett-Packard Co. 7301 No. Shadeland Avenue P.O. Box 50807 INDIANAPOLIS, IN 46250 Tel: (317) 842-1000 A,CH,CM,CS,E,MS

## lowa

Hewlett-Packard Co. 1776 22nd Street, Suite 1 WEST DES MOINES, IA 50265 Tel: (515) 224-1435 CH.MS\*\*

#### Kansas Hewlett-Packard Co. 7804 East Funston Road, #203 WICHITA, KS 67207 Tel: (316) 684-8491 CH

Kentucky

Hewlett-Packard Co. 10300 Linn Station Road, #100 LOUISVILLE, KY 40223 Tel: (502) 426-0100 A.CH.CS.MS

#### Louisiana

Hewlett-Packard Co. 160 James Drive East ST. ROSE, LA 70087 P.O. Box 1449 KENNER, LA 70063 Tel: (504) 467-4100 A,CH,CS,E,MS

## Maryland

Hewiett-Packard Co. 3701 Koppers Street BALTIMORE, MD 21227 Tel: (301) 644-5800 Telex: 710-862-1943 A,CH,CM,CS,E,MS Hewiett-Packard Co.

2 Choke Cherry Road ROCKVILLE, MD 20850 Tel: (301) 948-6370 A,CH,CM,CS,E,MP

## Massachusetts

Hewlett-Packard Co. 1775 Minuteman Road ANDOVER, MA 01810 Tet: (617) 682-1500 A,C,CH,CS,CM,E,MP,P\* Hewlett-Packard Co. 32 Hartwell Avenue LEXINGTON, MA 02173 Tet: (617) 861-8960

## CH,CS,E Michigan

Hewiett-Packard Co. 4326 Cascade Road S.E. GRAND RAPIDS, MI 49506 Tel: (616) 957-1970 CH,CS,MS Hewiett-Packard Co. 1771 W. Big Beaver Road

**TROY, MI 48084** Tel: (313) 643-6474 CH,CS

#### Minnesota

Hewlett-Packard Co. 2025 W. Larpenteur Ave. ST. PAUL, MN 55113 Tel: (612) 644-1100 A,CH,CM,CS,E,MP

## Missouri

Hewlett-Packard Co. 11131 Colorado Avenue KANSAS CITY, MO 64137 Tel: (816) 763-8000 A,CH,CM,CS,E,MS Hewlett-Packard Co. 13001 Hollenberg Drive BRIDGETON, MO 63044 Tel: (314) 344-5100 A,CH,CS,E,MP





# SALES & SUPPORT OFFICES

# Arranged alphabetically by country

## UNITED STATES (Cont'd)

Nebraska

Hewlett-Packard 10824 Old Mill Fid., Suite 3 OMAHA, NE 68154 Tel: (402) 334-1813 CM,MS

New Jersey Hewlett-Packard Co. 120 W. Century Road PARAMUS, NJ 07652 Tel: (201) 265-5000 A,CH,CM,CS,E,IMP

Hewlett-Packard Co. 60 New England Av. West PISCATAWAY, NJ 08854 Tel: (201) 981-1199 A,CH,CM,CS,E

New Mexico Hewlett-Packard Co. 11300 Lomas Blvd.,N.E. P.O. Box 11634 ALBUQUERQUE, NM 87112 Tel: (505) 292-1330 CH,CS.E.MS

New York Hewlett-Packard Co. 5 Computer Drive South ALBANY, NY 12205 Tel: (518) 458-1550 A,CH,E,MS Hewlett-Packard Co. 9600 Main Street P.O. Box AC CLARENCE, NY 14031 Tel: (716) 759-8621 CH Hewlett-Packard Co.

200 Cross Keys Office Park FAIRPORT, NY 14450 Tel: (716) 223-9950 CH,CM,CS,E,MS Hewlett-Packard Co.

7641 Henry Clay Blvd. LIVERPOOL, NY 13088 Tel: (315) 451-1820 A,CH,CM,E,MS

Hewlett-Packard Co. No. 1 Pennsylvania Plaza 55th Floor 34th Street & 8th Avenue MANHATTAN NY 10119 Tel: (212) 971-0800 CH,CS,E\*,M\*

Hewlett-Packard Co. 250 Westchester Avenue WHITE PLAINS, NY 10604 Tel: (914) 684-6100 CM,CH,CS,E

Hewlett-Packard Co. 3 Crossways Park West WOODBURY, NY 11797 Tel: (516) 921-0300 A,CH,CM,CS,E,MS

North Carolina Hewlett-Packard Co. 5605 Roanne Way P.O. Box 26500 GREENSBORO, NC 27420 Tel: (919) 852-1800 A,CH,CM,CS,E,MS Ohio Hewleti-Packard Co. 9920 Carver Road CINCINNATI, OH 45242 Tel: (513) 891-9870 CH,CS,MS Howlet Reckard Co.

Hewlett-Packard Co. 16500 Sprague Road CLEVELAND, OH 44 130 Tel: (216) 243-7300 A,CH,CM,CS,E,MS Hewlett-Packard Co. 962 Crupper Ave. COLUMBUS, OH 43229 Tel: (614) 436-1041 Eff: Nov. 25, 1983 675 Brooksedge Blvd. WESTERVILLE, OH 43081 CH,CM,CS,E Hewlett-Packard Co. 330 Progress Rd. DAYTON, OH 45449 Tel: (513) 859-8202 A.CH.CM.E\*.MS

Oktahoma Hewlett-Packard Co. 304 N. Meridian, Suite A P.O. Box 75609 OKLAHOMA CITY, OK 73147 Tel: (405) 946-9499 A\*,CH,E\*,MS Hewlett-Packard Co. 3840 S. 103rd E. Avenue, #100 P.O. Box 35747 TULSA, OK 74153 Tel: (918) 665-3300 A\*\*,CH,CS,M\*

Oregon Hewlett-Packard Co. 9255 S. W. Pioneer Court P.O. Box 328 WILSONVILLE, OR 97070 Tel: (503) 682-8000 A,CH,CS,E\*,MS

Pennsylvania Hewlett-Packard Co. 111 Zeta Drive PITTSBURGH, PA 15238 Tel: (412) 782-0400 A,CH,CS,E,MP Hewlett-Packard Co. 2750 Monroe Boulevard P.O. Box 713 VALLEY FORGE, PA 19482 Tel: (215) 666-9000 A,CH,CM,E,M

South Carolina Hewlett-Packard Co. Brookside Park, Suite 122 1 Harbison Way P.O. Box 21708 COLUMBIA, SC 29221 Tel: (803) 732-0400 CH,E,MS Hewlett-Packard Co. Koger Executive Center Chesterfield Bldg., Suite 124 GREENVILLE, SC 29615 Tel: (803) 297-4120

Tennessee Hewlett-Packard Co. 224 Peters Road, Suite 102 P.O. Box 22490 KNOXVILLE, TN 37922

Tel: (615) 691-2371 A\*,CH,MS Hewlett-Packard Co. 3070 Directors Row MEMPHIS, TN 38131 Tel: (901) 346-8370 A,CH,MS

#### Texas

Hewlett-Packard Co. 4171 North Mesa Suite C-110 EL PASO, TX 79902 Tel: (915) 533-3555 CH,E\*,MS\* Hewlett-Packard Co. 10535 Harwin Drive P.O. Box 42816 HOUSTON, TX 77042 Tel: (713) 776-6400 A,CH,CM,CS,E,MP Hewlett-Packard Co. 930 E. Campbell Rd. P.O. Box 1270 RICHARDSON, TX 75080 Tel: (214) 231-6101 A,CH,CM,CS,E,MP Hewlett-Packard Co. 1020 Central Parkway South P.O. Box 32993 SAN ANTONIO, TX 78216 Tel: (512) 494-9336 CH,CS,E,MS

Utah

Hewlett-Packard Co. 3530 W. 2100 South SALT LAKE CITY, UT 84119 Tel: (801) 974-1700 A,CH,CS,E,MS

Virginia

Hewlett-Packard Co. 4305 Cox Road GLEN ALLEN, VA 23060 P.O. Box 9669 RICHMOND, VA 23228 Tel: (804) 747-7750 A,CH,CS,E,MS

Washington Hewlett-Packard Co. 15815 S.E. 37th Street BELLEVUE, WA 98006 Tel: (206) 643-4000 A,CH,CM,CS,E,MP Hewlett-Packard Co. Suite A 708 North Argonne Road SPOKANE, WA 99212 Tel: (509) 922-7000 CH,CS West Virginia

Hewlett-Packard Co. 4604 MacCorkle Ave. P.O. Box 4297 CHARLESTON, WV 25304 Tel: (304) 925-0492 A,MS

Wisconsin Hewlett-Packard Co. 150 S. Sunny Slope Road BROOKFIELD, WI 53005 Tel: (414) 784-8800 A,CH,CS,E\*,MP

## URUGUAY

Pablo Ferrando S.A.C. e I. Avenida Italia 2877 Casilia de Correo 370 MONTEVIDEO Tel: 80-2586 Telex: Public Boolh 901 A.CM.E.M

## VENEZUELA

Hewlett-Packard de Venezuela C.A. **3RA Transversal Los Ruices Norte** Edificio Segre 1, 2 & 3 Apartado 50933 CARACAS 1071 Tel: 239-4133 Telex: 251046 HEWPACK A,CH,CS,E,MS,P Hewlett-Packard de Venezuela C.A. Calle-72-Entre 3H y 3Y, No. 3H-40 Edificio Ada-Evelyn, Local B Apartado 2646 4001, MARACAIBO, Estado Zulia Tel: (061) 80.304 C.E\* Hewlett-Packard de Venezuela C.A.

Calle Vargas Rondon Edificio Seguros Carabobo, Piso 10 VALENCIA Tel: (041) 51 385 CH,CS,P *Bioelectronica Medica C.A.* 

Calle Buen Pastor Edif. Cota Mil-Piso 2 y Semi Sotano 1 Boleita Norte Apartado 50710 CARACAS 1050A Tel: 239 84 41 Tele:: 26518

## ZIMBABWE

Field Technical Sales 45 Kelvin Road, North P.B. 3458 SALISBURY Tel: 705 231 Telex: 4-122 RH C.E.M.P

July 1983 Indicates main office 5952-6900

HP distributors are printed in italics.

.



MANUAL PART NO. 92862-90001 Printed in U.S.A. E0584 HEWLETT - PACKARD COMPANY Engineering Productivity Division 19420 Homestead Road Cupertino, California 95014